

Review

Not peer-reviewed version

Challenges to Water Resource Management—Role of Economic and Modeling Approaches

Ariel Dinar

Posted Date: 18 January 2024

doi: 10.20944/preprints202401.1354.v1

Keywords: Water; economics; globalization; climate change; competition; conflict; environment; valuation; CGE; game theory; experimental economics; institutional economics; international water; hydro-economic modeling



Preprints.org is a free multidiscipline platform providing preprint service that is dedicated to making early versions of research outputs permanently available and citable. Preprints posted at Preprints.org appear in Web of Science, Crossref, Google Scholar, Scilit, Europe PMC.

Copyright: This is an open access article distributed under the Creative Commons Attribution License which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited.

Disclaimer/Publisher's Note: The statements, opinions, and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions, or products referred to in the content.

Remiero

Challenges to Water Resource Management—Role of Economic and Modeling Approaches

Ariel Dinar

School of Public Policy, University of California, Riverside, USA, adinar@ucr.edu

Abstract: The field of water management is continually changing. Water has been subject to external shocks in the form of climate change and globalization. Analysis of water management is subject to disciplinary developments and inter-disciplinary interactions. Are these developments well documented in the literature? Initial observations on interdisciplinary literature suggest that results are fragmented, implying that a state-of-the-art review is needed. The objective of this paper is to close this gap by reviewing recent developments in water economics that address the increasing perceptions of water scarcity by looking first at changes in supply and quality of water, and then at impacts of climate change on water extremes. Among responses to such challenges, the paper identifies changes to water use patterns by including and co-managing water from different sources—surface and groundwater, wastewater, and desalinated water. Technological advancements also are among the resources that address water challenges. Water challenges reflect also on management of internationally shared water. A recent surge in scientific work identified international treaties as playing a significant role in water management. The paper reviews recently employed economic tools, such as experimental economics, game theory, institutional economics, and valuation methods. And finally, it explores modeling approaches, including hydro-economic and computable general equilibrium models that are being used to deal with water challenges.

Keywords: Water; economics; globalization; climate change; competition; conflict; environment; valuation; CGE; game theory; experimental economics; institutional economics; international water; hydro-economic modeling

JEL Codes: Q25; Q52; Q54; Q55; Q57

1. Introduction - Water shocks and future projections of water availability, quality, and use

A closer look at the conditions facing the water sector in many countries around the world makes the observer realize that major changes have been taking place in the supply patterns, the demand configurations, and use arrangements. For example, using estimates of global renewable water supplies [1–5], (Table 1.1: [6]) present changes to continental and global available renewable water resources. Using the global figures, mean global available renewable water resources are 42,780 Km³ and range between 39,780 and 44,750 km³ in dry and wet years, respectively. Data in [4] provides estimated and forecasted global water withdrawals that show an exponential increase between the years 1900 and 2015 rising from 750 km³/year in 1900 to 5,200 km³/year in 2015. While the available renewable water resources are fairly consistent, over time the need for water for different uses will increase due to population growth and changes in consumer taste and might be marginally decreased due to technological advancements and institutions. The net effect is that much of the world's total renewable water resources per capita are declining over time in a hyperbolic manner. Due to intensive development and industrialization, the level of pollution of water resources will increase and water will become less available for various types of consumption.¹ Challenges to these forecasts have been introduced recently in different publications, such as by Boretti and Rosa [7] by developing and

¹ For example, increased salinity in irrigation water makes it less adequate for irrigation of certain crops, and increased nitrate load in groundwater from agricultural pollution and fertilizer use makes groundwater use risky for residential consumption.



applying global models that account also for the effect of climate change and not only on population growth. Another set of considerations include alterations to water demand, which are the results of increase in living standards, and adaptation to water scarcity by technology development and institutions (both of which may not be well functioning in all countries). In a different approach a non-parametric framework was applied to analyze seasonal hydroclimatic regimes by classifying global land regions into nine regimes [8]. These regimes are used to assess implications for water availability due to concomitant changes in mean and seasonal precipitation and evaporation changes. Concluding the future water availability would drastically change by regions of the world. Using continental population and surface water availability, the per capita water availability in the world and its six continents between 1950-2080 is calculated and presented in [9]. They suggest that it is the local (regional) policies that consider societal preferences and priorities for allocation of available water for food production, residential consumption and environmental usage.

Where does all that lead to? In the next subsections we review the highly dynamic nature of the water sector and the interactions it realizes from global and local conditions. Later in the paper we will refer to the dynamic nature of water resource management being challenged by these changing conditions, and their need to adjust to using new economic and institutional approaches.

What does economics have to offer? Why is it important that economic considerations are part of the solutions? Does water play a role in growth and development of nations ([10–12])? As questioned in [11,12] would an increase in water scarcity affect the economic growth of countries? Theoretical and empirical analyses have an inverted-U relationship with economic growth and the rate of water use. Comparative analysis shows little evidence of severe diminishing returns to water allocated to production, thus resulting in falling income per capita. All it suggests is that the claims of some hydrological-based studies of a widespread global "water crisis" should be taken with caution. While there is strong evidence that rainfall variability and water availability have significant impacts on certain sectors such as agriculture, [10] suggests evidence of its effect on economic growth and other measures of aggregate economic activity remain inconclusive.

The water sector faces multiple challenges and objectives, including increased water use efficiency and improved conservation, land and other resource conservation, improved food security, and regulated pollution of water bodies from water use in the irrigation and domestic sectors. To address such challenges and achieve the objectives, the water sector realizes different policy interventions, including institutional reforms (e.g., water rights, water trade), pricing, quotas, taxes, and subsidies [6].

Economic models provide a useful framework for valuing water and its allocation among different uses. They assess the economic costs and benefits of different water uses and employ this information to guide decisions about water allocation and pricing. Approaches such as hydrological and water allocation modelling can provide insights into the dynamics of water resources management and help forecast water availability under different climatic and institutional scenarios. These models can be used to inform decisions about water allocation, infrastructure investments, and policies to address water scarcity and pollution.

Several recent papers provide reviews of the economic role in dealing with water challenges. Work in [13] surveys the literature on the economics of water scarcity and water demand. It provides information on demand for water in urban, agricultural, industrial, and demand for instream water uses. The review provides an overview of what is known about efficient water pricing, water allocation, and water trading. The paper highlights particular water management issues, such as allocation, to which economics has made important contributions, and areas in which additional research is needed.

A different perspective, suggesting to rethink water economics in light of such challenges is included in [14]. The authors observe that privatization, pricing, and property rights—the conventional economic policy interventions—have not performed well due to their failure to address adequately political economy considerations, including distributional conflicts. The paper identifies particular social and physical characteristics of water supply and demand and discusses their implications for three major aspects of water policy: infrastructure finance, water pricing, and property rights reforms. All these aspects show a major concern in sustainable management of water resources under uncertain climate change conditions.

A very useful review in [15] focuses on the role of economic analysis for helping to guide infrastructure investment choices, and for comparing the effectiveness of non-structural regulation approaches to water management. Among the economic approaches reviewed are optimization hydro-economic models (HEM), economy wide water-economy models (WEMs), socio-hydrological models (SHMs), spreadsheet-based partial equilibrium cost-benefit models, and others. The paper highlights recent cutting-edge works with WEMs and spreadsheet-based Cost-Benefit Analysis (CBA) models. Other advancements include multicriteria decision analysis, and game-theoretic modeling of noncooperative water institutions.

Under such changing conditions, it is important to compare available sets of traditional and new economic tools and approaches that could be put into action in order to address increased levels of scarcity, water quality deterioration, and water conflicts, to name a few challenges. This paper focuses on a review of advancements in economics of water management in recently published literature. The next section describes the methodology of selecting the publications to be included in this review. Section 3 reviews the various physical changes affecting the water sector in the recent two decades, which are reflected in the number and nature of the reviewed papers. Section 4 shows progress in water management approaches for dealing with local and international water uses. Section 5 reviews progress in management of international water. Section 6 presents specific economic tools and their contribution to dealing with water issues. It reviews experimental economics, game theory, institutional economic approaches, valuation approaches, and modeling approaches, such as hydroeconomic modeling and computable general equilibrium models. In section 7 all these advances are linked and assessed for their joint ability to address future challenges in water resource management. Finally, section 8 discusses unaddressed issues and agenda for future research.

While previous published reviews have focused on either sub-disciplines, or specific issues, this review provides a wide coverage of disciplines and issues. This paper calls for the adoption of an interdisciplinary approach to solving water issues; however, it has its own limitations that should be shared with the reader. The review addresses the role of economics in assessing different types of interventions, such as institutional reforms, taxes and subsidies, and support of water-saving technology adoption. The objective of the paper is to highlight traditional and new economic tools and modeling approaches that could address increased levels of scarcity, water quality deterioration, and water conflicts.

The review includes more than 200 citations of publications mainly from the past 20 years. However, it is obvious that many important and relevant publications may have been missed and do not appear in the reference list.

2. Materials and Methods

The methodology used to identify papers to be included in the review is based on both keywords search and personal expertise. Given the relatively long lasting experience of the author in the field of water economics, a structured outline of the review paper was first designed, helping with the topics of papers to be reviewed. In addition to this first step, a list of keywords was constructed including the major topics to be addressed in the review paper, which follow the list of advancements in water economics that have been experienced in the past two decades (e.g., water and CGE, water and Experimental economics, water and game theory, water and valuation economics, etc...). The search was conducted in datasets such as Google scholar and JSTOR. In addition, references in each paper that was included in the dataset were also searched and considered.

A total of 756 papers and reports were found and held for further relevancy evaluation (including working papers, and book chapters). All records were read by the author. This first screening yielded a workable dataset of 299 publications. The 451 publications that were removed from the initial dataset were less comprehensive in terms of the economic implications of the issues and methodologies applied. The remaining 299 publications were placed in bins that follow the subsections in the paper structure. An additional elimination of papers from the dataset was done by removing works that were found less relevant, based on a complete content review. The final usable dataset consists of 213 publications, most of which have been published during 2000 - 2024, but few also from years prior to 2000.

3. Physical changes faced by and introduced by the water users

As indicated earlier, water users have been facing major changes in terms of quantity, quality, and extremes to the supply of water. As a result, the way water has been managed had to be reconsidered. The following sections identify all observed changes and realize their impact on the

tools needed for appropriate water management under the new water reality.

3.1. Changes to water scarcity levels

Water scarcity is lack of a source of freshwater resources to meet the demands of water users for consumption, irrigation, and other uses. It can result from quantity shortage or access problems from water suppliers, or it may be due to water quality problems. Water scarcity affects people on all continents, and water use is increasing globally and has surpassed the rate of population increase by more than two-fold in the last century. Such unprecedented increase in water consumption has many regions facing water scarcity [16].

An interesting evaluation of the economic impact of water scarcity in various river basins around the world, implementing a Global Change Analysis Model (GCAM) is provided in [17]. GCAM is a hydro-economic model that decomposes the world into 32 geopolitical regions, 384 land-use regions, and 235 water basins. The model includes joined representations of the Earth's climate, economic, hydrologic, land-use, and energy systems in each region. Water scarcity is measured in the model as physical water scarcity according to the water withdrawal-to-availability ratio. Using several climatic scenarios, the GCAM suggests various interesting outcomes of the effect of exogenous impacts of water scarcity due to climate change in various river basins. While we are used to a negative effect of water scarcity on economic welfare, results suggest that, depending on the institutional and economic setups, positive basin-level impact from global scarcity can be realized in certain basins that demonstrate a comparative advantage over others. This comparative advantage can be realized if trade allows that basin to become a virtual water exporter through inter-basin trade and export of water-embedded goods to other basins. Such foundations of the hydro-economic models and their important role in addressing policies responding to water scarcity will be further discussed in a special section in the paper.

3.2. Changes to water quality

The quality of water resources in the world (both surface and groundwater) deteriorates over time, which presents a major challenge for policymakers in the water sector [18–21]. The authors in [18] review the types of contaminants, the type of water vulnerable to contamination, and the effects of such contaminants on human health. Contamination of surface water and groundwater from agricultural pollution of aquifers' water and waterways, industrial pollution of water resources, and elevated levels of household chemicals in wastewater have increased over the years. In the case of irrigated agriculture this is due to the needs of farmers to compensate for water scarcity, uncertainty in water supply, and increased effects of climate change on agriculture, such as elevated temperature levels that necessitate compensation with more fertilizers. Increased contamination levels in drinking water from different sources, resulting from loose regulations, has a direct effect on the health of populations consuming such water.² Regulation of these negative health and damaging effects vary and have different levels of effectiveness [22].

Mateo-Sagasta et al. [20] describe the extent of various pollution sources (nutrients, pesticides, salts, sediments, and organic matter) in irrigated agriculture and the possible policy interventions (regulatory instruments—standards; economic instruments—pollution taxes); education and awareness campaigns; cooperative agreements—tradeable pollution permits to address them [23]. In addition, Quinn et al. [24] examine a set of decision support models to address water quality management in real time, in different countries and sectors.

Lall et al. [21] focus on the deteriorating quality of groundwater globally. Pollutants from agriculture, industry, mining, energy production, and legacy landfills are an increasing threat to

² Similarly, contamination of groundwater and surface water affect the performance of the water-dependent ecosystems. Increased levels of various contaminants in irrigation water (including wastewater for irrigation) may find itself also affecting consumers through the food they consume.

groundwater. The authors suggest advances in modeling and data collection techniques (e.g., satellites) to support better management of water quality in aquifers around the world.

3.3. Impacts of external shocks (droughts, floods)

With climate change intensifying and altering the temporal and spatial distributional patterns of different water resources, it is anticipated that frequency and intensity of extreme hydrological events, such as heavy rainfall, floods, and droughts will increase. These extreme hydrological events may result in catastrophic effects on individuals (deaths) and communities (destruction of infrastructure). Economic evaluation of such effects and the possible approaches to manage natural resources under catastrophic events have been studied in [25,26] who examined the theoretical foundations for dealing with any catastrophic event, including droughts, and by in [27], who focused only on drought impacts assessment and handling.

3.3.1. Droughts

Drought differs from other disasters, and especially water-related disasters, in that it is a continuous phenomenon rather than having a short effect that strikes in a specific location (e.g., wind, flooding). Drought is a complex natural hazard since it doesn't have a universal yardstick. Drought conditions may vary, depending on the region [28]. Different definitions of drought reasoning include meteorological, hydrological, agricultural, and socioeconomic [28,29]). In addition, the different types of drought impacts are direct, indirect, and intangible. Each type (or combination of impacts) requires certain estimation methods and different policy intervention tools. Droughts not only affect the agricultural sector but also regions beyond the drought-stricken area, and sometimes the entire economy [30]. In many cases the indirect impacts of droughts could be more substantial than the direct impacts, affecting other industries and production sectors and seen along supply chains [31]. For example, drought can affect hydropower generation and, through loss of electricity supply, the production of food and other input to various production sectors of the economy, leading to a significant impact on the country's gross domestic product [32].

Several analytical frameworks have been suggested to address measuring of economic impacts of drought. Freire-González et al. [33] propose a conceptual framework based upon two sources of economic impact: "green water" and "blue water," arguing that since each source impacts the economy in different ways, they must be differentiated in any assessment of economic impact.

Additional studies refer to the impact of climate change on the severity and frequency of droughts, and the resulting economic losses. Naumann et al. [34] estimate that with no climate mitigation or adaptation, drought-related losses in the European Union and United Kingdom combined could reach more than €65 billion per year by the end of the century, compared with €9 billion per year at present. In a different focus, Gil et al. [31] measure direct and indirect effects of drought. They use a direct attribution model to measure direct effects, and a nested indirect attribution model for the indirect effects. The transmission of additional water scarcity effects from agricultural production to macroeconomic variables is measured through chained elasticities. While ample literature on drought effects on rural regions provides a good understanding of the social effects of this disaster, fewer studies have been focused on the urban sector. One example is the work in [35] who use monthly labor force data from 78 cities in Latin America to show the negative effects of droughts on the probability of urban employment and income. The authors find that the negative impact of droughts is greater than the impacts of flood events affecting these cities. Another line of works estimates the role of policy interventions to reduce the negative impact of droughts. For example, Booker et al. [36] applied a basin-level model to the upper part of the Rio Grande Basin to test whether institutional adjustments can reduce damages caused by drought. Findings suggest that future drought damages could be reduced by 20% and 33% per year through intra- and interstate water markets, respectively, that would allow water transfers across water-using regions affected differently by the drought.

3.3.2. Floods

Several studies were published recently that address the economics of flooding. Flooding is the "opposite" side of extreme effects of climate change and differs from droughts by having a more

targeted and short impact duration. Allaire [37] provides a review of economics of flooding in recently published work.

Sun et al. [38] argue that in addition to the significant flood damage to an area, such as effects on the local economy, disrupted transportation, and damaged infrastructure longer-term impacts and recovery occur within that area and the interaction between neighboring areas during the recovery process. The authors estimate the short- and long-term impacts of major and minor flood events on employment and business initiations in different sectors of the economy. Their findings indicate negative short-run impact on agricultural services and particularly small establishments, and positive impacts in the service sector. Tonn and Czajkowski [39] evaluate the inland flood risk in fluvial and pluvial plains from water ponding and insufficient drainage in urban areas. While pluvial and fluvial flooding are sometimes lumped together for purposes of modeling, insurance, and risk management, the authors treat them separately due to key differences that exist in the damage associated with these types of inland flooding. Floods originated from tropical cyclones have a higher percentage of pluvial flood insurance claims and higher average pluvial damage than non-tropical cyclone floods. For example, analyses of 2017's Hurricane Harvey claims show that zip codes with higher-than-expected losses tend to have high percentages of pluvial claims. Pluvial flooding and its local nuances exhibit the potential to reduce the accuracy of catastrophe models of expected flood damage. An interesting aspect of flood damage is its indirect impact of daily activity of the affected population. Ahmed et al. [40] identify short- to long-term adverse effects of the 2010 colossal flood on educational outcomes of children and adolescents in the flooded districts of Pakistan. They use a difference-in-differences approach to estimate the impact of the flood on educational outcomes by using household surveys' dataset to compare between flooded and non-flooded households before, during, and after the flood. The educational outcomes of children and adolescents in flooded households in rural areas compared with their peers in non-flooded districts were severely affected by the flood. And finally, Allaire [41] identified effective strategies for mitigating and coping with losses from flood. She compared two intervention policies—subsidized insurance and ex post compensation—in a theoretical model applied to the case of flooding in Bangkok. Findings suggest that an intervention to increase uptake of subsidized flood insurance does not deliver net social benefits relative to the status quo compensation program.

3.4. Changes in use of water sources (surface water, groundwater, treated wastewater, desalinated water)

Market conditions, such as changes in relative prices of goods or inputs, and changes in consumer tastes and preferences may lead to changes in use of the type of water for certain economic activities (irrigation or residential consumption). Water resources with potential for trade off, depending on prices and qualities, include surface water, groundwater, treated wastewater, and desalinated water. Several studies dealt with single water sources use while other studies focus on the opportunities with conjunctive use of various types of water, such as surface water and groundwater, surface water and treated wastewater, and even a conjunctive use of all types of water.

Esteban [42] reviews the challenges faced by groundwater (GW), including climate change, role in supporting GW-dependent ecosystems, effectiveness of regulatory interventions, and impact on land subsidence. Indeed GW is a resource threatened by both malmanagement of aquifers and by external shocks due to climate change [43–45]. With a 'future as usual' behavior of GW management, it is most likely that many locations on earth would face extreme water scarcity [46] and damages from land subsidence caused by long-term pumping of groundwater [47].

In addition to studies that estimate status of groundwater aquifers and changes in their water stock over time, other studies focus on behavioral consequences of users, and the effectiveness of policy interventions [48,49] for the case of groundwater-dependent ecosystems.

A large body of the literature addresses groundwater management options and their advantages and disadvantages. Esteban and Dinar [50] evaluated the ability of cooperative management of an aquifer to deal with over pumping and damaging groundwater-dependent ecosystems, while Molle and Closas [51] evaluated co-management options of an aquifer. Some of the works in this group evaluated policy intervention means. Esteban and Dinar [52] modeled effects of packaging and sequencing of pricing and quota interventions on both stock of the GW and the welfare of the users. Several works referred to the role of environmental flows services from groundwater. Pereau et al. [53] compared under various conditions the outcomes of optimal control approach (environmental

flows are introduced as an externality) with the outcome of an approach, in which environmental flows are modelled as a constraint to be satisfied. An interesting analysis of reduction of GW loss due to water uptake by an invasive tree is provided in Pongkijvorasin et al. [54], who provided decision rules to trigger the removal of the invasive tree. A recently gaining popularity is the technology of managed aquifer recharge (MAR) that allows the intended recharge of aquifers with different types of water. In developing an MAR model for a region in California, Reznik et al. [55] demonstrate that intentional recharge is of high benefit to the region, potentially increasing average groundwater levels in the region by 20% over a 20-year horizon.

As suggested and proven by Dinar and Tsur [6], different types of water and different waterusing sectors should be considered jointly for handling water issues and for finding socially acceptable solutions to water issues. Following are several examples for the conjunctive use of surface water, groundwater, and treated wastewater. Economists have long advocated for integrating treated wastewater in irrigated agriculture to achieve both surface water conservation and reduce environmental pollution from disposal of residential sewage to the environment. Reznik et al. [56] develop a regional multi-sectoral model of water quantity-quality interaction among the urban, agricultural, and environmental sectors. Highlighting the feasibility of reuse, they formally construct sufficient conditions that support the superiority of infrastructure development and conveyance of treated wastewater for irrigation, when measured against other common disposal alternatives. In a more empirically related work, Reznik and Dinar [57] apply the framework in [56] to a particular region in California accounting for uncertainties in the availability of natural water resources and regulatory constraints concerning wastewater discharge. While they find reuse for avocado crop irrigation unwarranted, still utilizing that practice to support the agricultural economy in the region is economically inexpensive and could be feasible under a set of circumstances that are most likely noticeable in the region. Wastewater treatment cannot be addressed without introducing considerations of climate change. Climate change does affect cost of wastewater treatment and its inclusion in future investment plans for wastewater treatment. Reznik et al. [58] find evidence of climate change impact on treatment costs, using a unique dataset from China. The analysis also simulates potential impact of future policy and climate scenarios on costs of treatment.

A resource that increases in importance in certain locations around the world is desalinated sea (or brackish) water. Not too many studies exist on the economics of desalination and the use of desalinated water for irrigation. Economic considerations are essential for dealing with desalinated water for any use (either for residential or for irrigated agriculture). Younos [59] provides an overview of factors that determine desalination cost, typical desalination cost estimation models, various cost factors, and approximate costs based on a review of case studies and available literature at that time. More location-relevant work is also reviewed. The desert countries in the Arab region, such as Saudi Arabia, initiated irrigation of crops with desalinated water. This has been seen at the early stages as an infeasible practice, but as water became scarcer, more and more of such practices are seen in the region. Sewilam and Nasr [60] describe the process of expanding desalinated water use for irrigated agriculture and the economics of such a practice in the Arab world. Kaner et al. [61] develop and apply a biological-physical model for crop response to salinity coupled with economic calculations of farm-based costs and benefits to determine the impact on yield from irrigating with desalinated water in Israel. The value of yield is then compared with the economic feasibility of investment in farm- or regional-scale desalination plants to supply high-quality water as an alternative to irrigation with brackish water that is available in the region. The results suggest that the predicted profit from production of high-value, salinity-sensitive crops irrigated with either pure desalinated or blended desalinated and locally available brackish water justifies the desalination cost for agriculture at present market prices.

Technology adoption and use in water-consuming economic activities has long been the center of addressing water scarcity and quality challenges. The next section focuses on economics of technology use to conserve water and sustain economic productivity.

4. Role of technological advancement in addressing water resource management challenges

Technological advancement is one of the strategies to cope with water quantity and quality challenges. Governments and the private sector invest in technology development and in their wide adoption by water users in the various sectors. The next three sections will address technical and

-

economic feasibility of water-saving and pollution-reduction technologies, and new studies on adoption of such technologies by water consumers in the agricultural and residential sector.

4.1. Feasibility of water-saving technologies and pricing

Studies on the feasibility of water-saving technologies are an important first step in the economic assessment of the possibility of new technologies and/or management practices to produce economically credible flows of benefits from water savings, while keeping production (in agriculture) or satisfaction from water usage (in residential uses) at a level that will lead to a positive benefit/cost ratio. Inman and Jeffrey [62] conducted a review of residential demand-side management (DSM) tools aimed to save water in various types of household under varying conditions. Tools include financial approaches (metering, pricing), technological tools (indoor, outdoor), educational approaches, operations and maintenance tools, and regulations. The paper also compares implementation costs and effectiveness of the demand-side management tools and approaches.

Several representative studies include review of the literature and several case studies, all providing methodological framework and evidence for the relevance of the approaches used. Perry and Steduto [63] found that very few examples exist of carefully documented water-saving impacts of hi-tech irrigation. Their conclusion is that introducing hi-tech irrigation in the absence of controls on water allocations will usually lead to increases in water consumption per unit area, increase in the irrigated area, and increase in the amount of water extracted and applied. Pérez-Blanco et al. [64] reach a similar conclusion, specifically that water conservation technologies should not be viewed as a tool for achieving water conservation, but rather as a means for stabilizing and increasing agricultural water productivity and farmers' income in regions facing water scarcity. Ward and Pulido-Velazquez [65] demonstrate by using an integrated basin-scale analysis linking biophysical, hydrologic, agronomic, economic, policy, and institutional dimensions of the Upper Rio Grande Basin that adoption of more efficient irrigation technologies reduces valuable return flows and limits aquifer recharge, hence increasing water scarcity. Similar finding, namely that higher efficiency rarely reduces water consumption are reached by Grafton et al. [66], who show that to mitigate global water scarcity, increases in irrigation efficiency must be accompanied by comprehensive water accounting and measurements, a cap on extractions, an assessment of uncertainties, valuation of trade-offs, and a better understanding of the incentives and behavior of irrigators. These findings are similar to those reached by Dinar and Zilberman [67] when evaluating economic consequences of resourceconserving, pollution-reducing technologies. By coining the concept "expansion effect" they introduced consideration beyond the field level and the interaction between water scarcity and water quality. Depending on the relative limiting factor (quantity-quality), they argue that modern (resource-conserving) technologies may provide incentives to conserve resources if combined with appropriate policies of input and output prices and regulations on pollution [67:pp346].

Several studies introduce benefit/cost analysis as to the financial/economic feasibility of water conservation irrigation systems under local conditions. Siderius et al. [68] develop a modified traditional cost curve approach with an improved estimation of demand and increasing marginal cost per water conservation while also correcting for impacts on downstream water availability. The framework is applied to three major water-stressed river basins—Indus, Ganges, and Brahmaputra. Results suggest that at basin the equilibrium price for water is too low to make the majority of water conservation measures cost effective. Vatta et al. [69] look at a monitoring device to assess the impact of tensiometer technology on the consumption of groundwater and electric power in paddy cultivation in Indian Punjab, and its subsequent economic benefits. Findings suggest that compared to the continuous flooding and furrow irrigation methods at present, the tensiometer-based application of irrigation water reduces water and power consumption without any yield reduction.

Table 1. Studies on water conservation and their findings.

Published article	Finding
	Resource-conserving technologies may provide incentives
[67]	to conserve resources if combined with appropriate policies
	of input and output prices and regulations on pollution.

[(0]	Relative effectiveness of different DSM water saving tools in		
[62]	the residential sector.		
	Adoption of more efficient irrigation technologies reduces		
[65]	valuable return flows and limits aquifer recharge, hence		
	increasing water scarcity.		
[63]	Introducing hi-tech irrigation in the absence of regulation on		
	water allocations will usually lead to increases in water		
	consumption per unit area, increase in the irrigated area,		
	and increase in the amount of water extracted and applied.		
[66]	Increases in irrigation efficiency must be accompanied by a		
	whole arsenal of regulations such as water measurements, a		
	cap on extractions, an assessment of uncertainties, valuation		
	of trade-offs.		
[69]	Compared to the continuous flooding and furrow irrigation		
	methods at present, the tensiometer-based application of		
	irrigation water reduces water and power consumption		
	without any yield reduction for rice irrigation.		
[64]	Water conservation technologies should not be viewed as a		
	tool for achieving water conservation, but rather as a means		
	for stabilizing and increasing agricultural water		
	productivity.		
[68]	At basin the equilibrium price for water is too low to make		
	the majority of water conservation measures cost effective.		

Another strand of the literature, mentioned here very briefly, has focused on the effectiveness of different tariff design to achieve water conservation and the tradeoffs between equity, economic efficiency, and cost recovery [70]. Using a hypothetical community served by a water utility, the authors analyze how moving from a uniform volumetric tariff to increasing block tariff (IBT) affects households' water use and water bills, and how such changes affect measures of equity and economic efficiency. Investigation of behavioral aspects of responses to water tariffs in a survey of the implementation [71] provides international evidence on using IBTs to conserve water, which is mixed. This highlights the operational challenges of implementing effective IBTs.

4.2. Feasibility of water pollution reduction technologies

Pollution reduction technologies could include residential wastewater treatment or irrigation-related chemical return flows pollution reduction technologies. Goffi et al. [72] compare the economic feasibility of 37 wastewater treatment technologies, considering cost-related indicators for selecting the "ideal" technology by fitting it to specific features in cities. Using the criteria of net present value and annualized net present value, they prioritize the evaluated systems. In the agricultural sector, leaching from fields of irrigated agricultural sector is often the largest source for aquifer recharge in semi-arid groundwater basins, but contamination from fertilizers and other agro-chemicals may degrade the quality of groundwater. Mayzelle et al. [73] compared the viability of two low-impact crops—alfalfa and vineyards—and new recharge basins as an alternative land use in recharge buffer zones around communities facing contaminated groundwater in the southern Central Valley of California. Buffer zones would maintain the economic integrity of the region and concur with prevailing regulations.

4.3. Adoption of new technologies (residential, agricultural)

Technological and economic feasibility of new technologies do not guarantee their implementation by water users. The following studies report the determinants of adoption of technologies that save water both in irrigated agriculture and by households, and adoption of technologies that reduce pollution.

A study by Schaible and Aillery [74] in the United States finds that despite technological innovations, at least half of U.S. irrigated cropland acreage is still irrigated with less-efficient, traditional irrigation application systems. Studies on adoption of irrigation technologies, such as the ones cited below may explain why. de Witt et al. [75] investigated personal barriers of farmers to a water-scheduling technology that was developed and provided free of charge by the government. The non-use of the free service is explained mainly by the time and cost associated with the technology's initial set-up, use, and interpretation of information by the individual farmers. Gui and Gou [76] explain regional differences in adoption of technologies to capture alternative water sources for different domestic uses. One of the important findings is that adoption of water technologies in rural and metropolitan areas differ and, thus, both technologies and distributional policies should be different. Given the high share of domestic water consumption for irrigated landscape (in Florida and California), [77] studied urban landscape water conservation innovation adoption in Florida to find major differences in speed of adoption among households in different societies. Results suggest opposite processes of diffusion of water-saving technologies by societies, which leads to a need for different dissemination strategies.

A different approach to measuring adoption of water-saving technologies has considered a process of adopting not one technology, but rather a bundle of technologies that benefit each other. For example, using drip irrigation could be made much more effective if a fertilization technology takes advantage of the precise water allocation via the drip system. In a study focused on avocado growers in California, [78] showed how a variety of irrigation technologies bundled with water management practices help growers through times of limited water supplies and elevated salinity levels. And yet another example of budling irrigation water sources was demonstrated in [79] that evaluates the benefits to irrigated agriculture in California from having access to multiple sources of water, each characterized by quantity, security (water right), and quality (salinity content). As in other bundled technology adoption, while lower-quality waters, less-reliable water, and less water all negatively impact agricultural land values, holding a water (bundle) portfolio positively impacts land values through its role in mitigating the negative aspects of these factors and reducing the sensitivity of agriculture to climate-related factors.

So far, the review referred to domestic water challenges and economics approaches to address them. The next section introduces an issue—international water—with rising importance in regional management of water resources.

5. Management of Internationally shared water

International river basins are water bodies that are shared among more than two riparian states. As of the 2018 update, 310 international river basins cover nearly 47% of the global land area [80]. With such a major mass of land, dealing with management of international water may have a wide impact globally, and is of interest to various disciplines dealing with international water, such as economics, engineering, political science, international relations, international law, and combinations of these disciplines. The literature on management of international water has seen much collaboration among scholars from these disciplines as was summarized in [81]and [82].

One of the most effective arrangements for dealing with international water is the treaty tool, which is used among riparian states to address different aspects of water management in the basin (e.g., allocation of the water, allocation of benefits and costs, regulations of pollution). The analysis of international treaties has gained momentum in the literature, allowing economists and political scientists to join forces in addressing research on the efficiency of such treaties. Recent developments in the roles of treaties for cooperative management of international water are discussed in [83]. Author says that treaties provide states with a platform for dealing with conflict at the basin level in addition to having the means to produce benefits for sustained cooperation in the basin.

Given the economic-political nature of cooperation in international river basins, joint work by economists, engineers, and political scientists is most adequate to address issues of water management on a basin scale. A sample of recent works that highlight different angles of the issues faced by riparian states follow. One of the obvious observations highlighted by many researchers is why there are so few agreements on international water. To answer such question, [84] developed an economic-political framework to explain the likelihood of treaty formation in international river basins with a different number of riparian states. They used the interaction between the transaction

cost theory, and the economies and diseconomies of scale theory to explain the likelihood of expanding a coalition of collaborator states in a large basin. Some important findings suggest that transaction costs increase the likelihood of treaties with environmental regulation, and greater economies of scale lead to fewer treaties with environmental regulation. Transaction costs raise the likelihood of treaties with water allocation, but economies of scale have the opposite effect. Transaction costs tend to reduce the number of issues included in the treaty, while economies of scale tend to increase it. Higher transaction costs tend to increase the number of institutional mechanisms incorporated in a treaty, and greater economies of scale increase the number of treaty institutions for managing the basin to be shared among the riparian states.

The findings confirm the results in [85], who focused on the impact of the negotiation context on the treaty content, as was already suggested by [86], and [87], who stressed the importance of the transaction costs of negotiating and sustaining treaties, along with the need to invest in institutional design to address transaction costs.

Global analyses of treaty performance was conducted in [88] for all bilateral river basins, and in [89–91], focusing on the effectiveness of water management during climate change on cooperation. The conclusions from all these studies show that well-designed and crafted treaties can create stable arrangements under extreme climatic conditions.

Another important aspect motivating cooperation among states has been identified in [92] in the case of differences in power, attitude, and capacity of riparian states with regard to the environment and environmental pollution. Asymmetries in concern for the environment, or the ability of the parties to abate pollution, are most likely the result of economic differences, and influence the willingness to pay for pollution abatement. While such differences could result in tensions, they can incentivize the states (most likely richer and more developed states) to encourage cooperation in the basin

A strand of research uses the concept of issue linkage to allow pulling the parties in a deadlock conflict to move to a cooperative solution. Several works [93,94] identified the theoretical advantages of issue linkage, and others demonstrated its application in the case of the conflict over the Mekong River Basin between China and the lower Mekong States—Laos, Cambodia, Viet Nam, and Thailand—[95–97].

Other works focus on the role technology plays in addressing conflict in river basins facing high water scarcity, via exchanges (linkages) between water and energy [98], and via production of new desalinated water in exchange for traditional surface water [99].

Cooperation also can be challenged by the allocation arrangements of incremental costs or benefits from cooperation to the participating parties. More on that can be found in section 4.2 (Game theory), which is part of a broader section that introduces economic tools such as experimental economics, game theory, institutional economics, valuation methods, and modeling approaches (such as hydro-economic modeling and computable general equilibrium modeling.

6. What do new economic tools and approaches have to offer?

Recent developments in software and computer capacity gave rise to use of several economic tools and approaches in analysis of integrated water resources management (IWRM), which was less apparent in previous decades. The increased use of experimental economics, game theory, institutional economics, valuation methods, and different modeling approaches, such as hydroeconomic (HEM) models, and computable general equilibrium (CGE) frameworks will be reviewed next.

6.1. Experimental economics

Experimental economics is a relatively new field that has been applied in recent years to issues related to policy evaluation for water management under conditions of scarcity, deteriorated quality, and external shocks such as climate change. A recent review of the role of studies applying experimental economics to evaluate policy reforms in the water sector [100] suggests that experimental economics, either by itself or in conjunction with other approaches, can save society the grievance of high social costs of failing policy implementation. Evaluation of policy reforms using experimental means can be performed in the lab or in the field prior to the actual implementation of the policy. The challenge faced by experimental economics as identified in [100] is the need to

consider how scaling up the proposed policy interventions can remain relevant and not create or exacerbate societal disparities and ethical challenges from the externality addressed. It included works on establishing water markets and conservation, water use and conservation regulation in different sectors, payment for water-related ecosystem services, and emissions trading to regulate water pollution reduction.

An earlier review of advantages and challenges in experimental economics applications to water sector issues [101] addresses the role of experiments in advancing economic understanding of water resource issues. The review included works focusing on self-regulating behaviors in common-pool resources (groundwater), the efficiency of alternative water quantity and quality markets, the comparative efficacy of alternative types of ambient pollution control instruments, and water conservation measures.

A hand-picked set of works presented in [100,101], and elsewhere are summarized in Table 2. While experimental economics is growing in coverage of water issues, the table below identifies five major issues for which experimental economics approaches are reported in this review, including water conservation, water markets, subsidy regulations, watershed management, and international water [102–113].

Published article	Policy focus	Issue addressed	
[102]	GW regulation		
[103]	Residential water	Water conservation	
[104]	Water Conservation regulation		
[105]	Smart water Markets	_	
[106]	Water markets and instream values	- TA7 (1 (
[107]	Water quality trading	- Water markets	
[108]	Agricultural subsidies to regulate water quality	-	
[109]	Electricity subsidies and GW level		
[110]	Electricity subsidies and GW level	Subsidy regulations	
[111]	Electricity subsidies and GW level		
[112]	Collective action for watershed management	Watershed mgt.	
[113]	Issue linkage	International water	

Table 2. Recently published works on water and experimental economics by issue.

Experimental economics has a significant role in the design and evaluation of policies in the water and environmental sectors, aimed at internalizing or mitigating effects of negative externalities. However, Banerjee [100] argues that experiments have to take into account that policies don't operate in a vacuum. Therefore, for an experimental design to be relevant, the context of the analysis is important. Context may include the institutional setting, the social and political settings, and the physical setting. All these considerations may have far reaching unintended consequences, even if the negative externality is internalized.

6.2. Game theory

Game theory approaches have been applied to deal with water resource management, given that water is a common pool resource subject to strategic behavior. As climate change is expected to result in water scarcity and deterioration of water quality around the world, water quantity and quality are becoming a major challenge for water managers and for nations that share limited water resources. The literature has seen an increase in published works on the use of various game theory frameworks at local regional and international setups. Publications such as [114–116] provide collections of studies that apply game theory models to various types of water issues at local, national, and international levels. A review of works that apply game theory models spanning the years 1940 through 2014 can be found in [82].

Game theory studies apply cooperative and non-cooperative frameworks and their relevance to various water management issues. Several of the studies refer to domestic conflict situations among sectors or regions [117,118], and others refer to water issues in international river basins ([119–127]). All of these works refer mainly to allocation of scarce water. Some works [124] develop basin-level

hydro-economic models to calculate the benefits and costs of the various basin riparian states. Others [121] calculate the coalitional benefits and costs outside of the game theory framework and then incorporate the calculated values into the game theory allocation equations.

To be more specific about the contributions of several of the studies cited in the previous section, [117] compare the outcomes of two analytical frameworks set to optimize scarce water allocation between agricultural production and environmental consumption. The two analytical frameworks—a cooperative game theory framework and a negotiated role-playing framework—were applied to a river basin in South Africa. The authors concluded that the role-playing negotiated framework has more flexibility to incorporate real world situations, and thus, to provide more options for the stakeholders to consider in their allocation challenge.

The works by [119,120] develop a basin-level game focusing on geographies that can affect the stability of the allocations. The former work suggests measures of equity among the participants in the simplified game while the latter work is more realistic in the sense that it reflects conditions in the Amu-Darya river basin and actual agreed allocations among the several riparian states under scenarios of high water scarcity that this basin faces.

Pham Do [128] identifies challenges facing game theory, as well as opportunities that game theory can address, to help water managers achieve sustainability, equity, and optimal conditions. An interesting part of the paper deals with distinguishing between cooperative and noncooperative game models and static, versus dynamic game models. I will refer only to the latter classification.

Both cooperative and noncooperative games can be classified as *static* or *dynamic*. They differ in the timing decisions that are made by the players. Participants in static games consider the immediate action of the opponent player, while players in dynamic games consider the responses by the opponent players following their own decision. In static games, all players choose their strategy at the same time. The static approach has been developed and applied to decision-making in WRM, particularly groundwater resource management [129–131]. Players in dynamic games can move with their strategy sequentially or repeatedly. Dynamic game models have been developed as both discrete and continuous. The dynamic approach (both theoretical and empirical analysis) has been recently applied to multipurpose water projects [132,133], and analysis of spatial groundwater management and policies [134].

In recent years we observe a surge in works applying game theory to situations with the impact of climate change on water resources [135]. The authors of this publication demonstrate the use of a bargaining game and the Nash bargaining solution with two methods: a symmetric, and an analytic hierarchy process (AHP) method in allocating water of higher level of scarcity to users. The AHP method provided better outcomes for players than the symmetric method. The results suggest that in the use of a cooperative bargaining game, water allocation can be achieved in a more efficient way.

In a different study focused on the Colorado river basin [136] the authors apply a bankruptcy game framework to allocating climate-induced, over-committed water rights agreements to competing stakeholders of different sectors in the Salton Sea region. They used two models for allocation: one involving a social planner approach that maximizes regional welfare, and the second focusing on the bankruptcy rules of proportional deficit (cutback) and constrained equal award. The findings suggest the proportional cutback framework to be less conducive to regional welfare, although it presents a more politically feasible and robust option.

Game theory modeling has demonstrated an ability to address cooperative and noncooperative aspects of water resource management situations. However, the ability of game theory to provide "operational" suggestions to water managers has been limited by the intensive computational needs of the algorithms developed. An example of such computational intensity needed for making the game theory results useful for managers can be found in [137]. The authors refer to a heterogeneous project both in terms of the landscape of the area under development and the participants (users), both of which lead to a more complicated set of cost allocation mechanisms than in homogeneous projects. The analysis presented in their paper uses cooperative game theory to develop schemes for sharing costs and revenues from a project involving various beneficiaries in an equitable and fair way. The proposed approach is applied to the West Delta Irrigation Project in Egypt. It utilizes a differential two-part tariff that reproduces the allocation of total project costs using the Shapley Value. The proposed differential tariff, applied to each land section in the project reflecting their landscape-

related costs, would be preferred by the users in the different land blocks of the project. It contrasts the unified tariff that was proposed, using the traditional methods in the project planning documents.

6.3. Institutional economics

The analysis of water institutions has been on the rise in recent years. Failing or malfunctioning water sectors have been challenged with a need to improve their institutional performance. Most of the published work is descriptive, attempting to explain why some institutional arrangements have worked in some places but failed in others. For example, [138] examines searching for panaceas in irrigated agriculture, such as coordinating institutions, and looking at policies promoting government agencies, user organizations, and water markets. Questions such as how to measure institutional success have been rarely answered. Most water institutional analyses have been conducted using the case study approach, so extrapolation from one location with certain conditions to another location with different conditions was impossible.

Decentralization is one of the institutional reform interventions attempted in recent years in many water systems, such as river basins. The literature has mainly described the process rather than explained its determinants. A quantification of a global institutional reform focusing on the process and performance of decentralization in river basins in 83 and 27 river basins around the world was performed in [139] and [140], respectively. The studies estimated the contribution of local conditions and supporting policies to the effectiveness and speed of the reform and identified the level of success as a function of conditions in each of the basins.

In two separate studies, [141] and [142], the authots apply quantitative models to decompose water institutions in 43 and 17 countries around the world, respectively, for a comparative analysis of performance of their water institutions. And [143] developed diagnostic models applied to 87 poor countries to find that water-related public services contribute to per capita GDP and to improved public health outcomes across these developing countries. The relationship between the wellness of the water institutions and the performance of the economy were also shown in [144]. Maria-Saleth and Dinar A. [145] applied the analytical framework they developed in [141] to quantitatively show the structural and functional linkages within institutional structure, and to indicate their performance implications and strategic importance for promoting institutional reforms. Bandaragoda [146] builds on the suggested framework in [141], (which was published in a couple of World Bank working papers and technical papers [147,148], and develops a qualitative framework for institutional analysis of water resources management in the context of a river basin.

To promote the important analysis of water institutions, Berbel et al.[149] dedicate a special issue in which they demonstrate the role of institutions during scarcity and drought situations. In a similar vein, authors in [150] quantify the significant factors of water institutions and their effect on the different aspects of water sector performance under scarcity and water quality challenges in India.

6.4. Valuation methods

Recent years have brought a significant increase in publications that focus on environmental valuation [151]. Some argue that this is the result of global change, affecting vulnerable water resources, but also increasing their attributes that can benefit different segments of society. Globalization has made it easier to move along the attribute spectrum due to advances in availability of technologies that can help alter water quality and make it usable for different purposes. Another reason for the increased importance of valuation methods is the objective of increased water efficiency by maximizing the value of the resource as water becomes scarcer. Environmental valuation has an important role in informing the development of standards for water used for different purposes, such as drinking, irrigation, and water-dependent ecosystems. In addition, environmental valuation can inform optimal investment in water treatment, such as the net benefits of desalination, treatment of recycled water, and managed aquifer recharge, to name a few.

For obtaining values associated with global changes to water demand, water supply and water quality, valuation methods need to rely also on non-market values. As such, valuation requires the use of either indirect valuation approaches (e.g., travel cost, hedonic pricing, defensive expenditures), or direct survey techniques (contingent valuation, choice experiments) [152,153].

The use of valuation methods has seen a strong boost with the unfortunate well-known Exxon Valdez oil spill on the shores of Alaska on March 24, 1989, causing some 11 million gallons of crude

oil to spill into the water [154]. Since then, valuation methods have been used to assess damages of other oil spills [155,156], as well as assessment of values of the environment resulting from development projects, such as [157], who conducted an ex ante benefit—cost assessment and forecast market-clearing prices and quantities for ecological infrastructure investment contracts in the Panama Canal Watershed. And [158] ,who assessed the extent of altruism vs. self interest in the valuation of community drinking water quality improvements investments in Canada. Another example is [159], who compared the contingent valuation and choice experiment methods in developing solid waste management programs in Macao.

Johnston et al. [160] propose contemporary best-practice approaches for stated preference (SP) studies. These recommendations consider the use of SP methods to estimate both use and non-use (passive-use) values, and cover the broad SP domain, including contingent valuation and discrete choice experiments. The paper focuses on applications to public goods in the context of the environment and human health.

Valuation methods are not innocent of problems and caveats. Some of the issues raised regarding their reliability and validity can be found in [161], who consider contingent valuation and travel cost approaches in their analysis. In a different study, authors [162] review previous literature and empirically show that presentation of the topic may matter in the case of contingent valuation and choice experiments, as broad descriptive terms may mask the many design and methodological differences seen in implementations of the approaches.

Valuation methods, such as the travel cost method and the choice experiment method, were also used in the case of cultural heritage [163], natural attractions [164], climate change hazards to coastal regions [165], and evaluation of services from the Rasmar-protected Cheimaditida wetland in Greece [166].

And finally, a review of water valuation metrics for the manufacturing sector is presented in [167]. The authors define the full value of water as price+true cost+other internal costs+indirect/opportunity costs, and compare results across 20 industries and 8 countries between 2005 and 2020. The authors show, using many international cases, that by extending the concept of water value to include various relevant cost and value components, such fuller water valuations the development and deployment of cost-effective water-conservation technologies more economical, thus, improving the sustainability of the manufacturing sector with respect to water.

6.5. Modeling approaches

6.5.1. Hydro-economic modeling

Hydroeconomic models (HEMs) help in analyzing water management problems by integrating models that explicitly represent interactions between the hydrology aspects of the water body analyzed and the economic, infrastructure, legal, and institutional aspects of the region under consideration. The combination of these aspects of management provides better-informed results for decision-making in the complex environment in which water management operates. HEMs allow both water supply and demands to be economically and hydrologically defined and described in [168]. Over the years, HEMs have been developed to address several important issues, such as adaptation to climate change, environmental flows, conflicts among domestic sectors, and international water, to name a few.

Several publications provide reviews of the body of literature on HEMs. Such publications include [169], which reviews 80 HEMs applied to water analysis in 23 countries. The paper identifies the key stages in model design, formulations, levels of integration, spatial and temporal scales, and solution techniques. Bekchanov et al. [170] classify reviewed HEMs into network-based (simulation or optimization) models and economywide (input-output or computable general equilibrium) models. The paper highlights the primary differences in the applications and interpretations obtained using these approaches. The paper suggests that additional efforts are needed to account for the range and complexity of linking water systems and society more realistically, particularly regarding ecology and water quality, and the food and energy sectors. A review integrated hydro-economic models that aim to capture the complexity of interactions between water and the economy can be found in [171]. The paper identifies issues and future research directions in integrated hydro-economic modelling, which are demonstrated using a variety of case study applications worldwide.

A different angle is presented at a review in [172], comparing the relevance of HEM approaches for large-scale decision-making in multi-sectoral and multi-regional river basins.

Daclin and Fernandes [173] propose modeling approach that identifies how to apply water management instruments integrated with each other to deliver a water allocation strategy reconciled with economic development projections and changing water use preferences.

A couple of publications demonstrate the use of HEMs to address transboundary river water sharing disputes [174,175] by introducing benefit-sharing arrangements to the analysis and developing a continental-level HEM [176].

Another aspect of water management addressed by HEMs is the handling of water pollution either by urban centers [177], or by the agricultural sector [178,179].

Environmental flows and impacts of water scarcity on ecosystems are a recent focus of HEMs. Several papers demonstrate this relatively new trend. Crespo et al. [180,181] develop an HEM to assess the value of environmental flow in the Ebro Basin in Spain. Levers et al. [182] develop an HEM to assess the effects of buying water for the environment in the Salton Sea in California. And Momblanch et al. [183] develop and apply an HEM that allows the use of ecosystem services to represent the environment.

Groundwater management is well addressed by HEMs, showing an ability to account for the open access of groundwater and their significant negative externalities [184,185]. The interactive relation between groundwater and a river are modeled by [186] in an HEM aimed at guiding sustainable basin management.

Sectoral integration is also one of the special features found in HEMs. Authors [187] evaluate integration policies to allocate water between agriculture and the urban sector.

Several papers demonstrate the use of HEMs for assessment of possible adaptation strategies to climate change. Esteve et al. [188] assess climate change impacts and adaptation in irrigated agriculture. Expósito et al. [189] review applications of HEMs to assess effectiveness of water policies under climate change at basin scale, and Kahil et al. [190] model with HEM the impact of water scarcity and droughts for policy adaptation to climate change.

6.5.2. Computable general equilibrium (CGE)

Because of their central role in both developing and developed economies, water resources are the focus of many intervention policies. Since agriculture consumes the lion's share (70–90%) of annual renewable fresh water on earth, it is a sufficient reason for policymakers to focus their efforts on improved performance of scarce water use in irrigated agriculture. Policies targeted at multiple objectives, including income transfer, food production security, environmental sustainability, and resource conservation may lead at times to pervasive outcomes at various interacting sectors. This system of cause and effect holds also for the urban water sector, as well as for the industrial and environmental sectors. Therefore, water as a policy target has to be regulated at the economy-wide level when being allocated among competing uses.

In recent years, we have witnessed increased globalization and climate change considerations, both of which strongly suggest that water policy is no longer a sectoral, or regional, but an economy-wide matter. Recognizing this trend gave rise to studies of an economy-wide nature. While many economy-wide analyses (mainly computable general equilibrium—CGE—models) have been published in the economic literature on water, little can be generalized, mainly because these studies use different assumptions and structures of the economy. For example, many CGE studies on water that have been reported in the literature treat irrigated agriculture as one sector or activity. Such structures are only appropriate in economies in which physical (soil quality, water availability, etc.), economic and social conditions (crop mixes, proximity to markets, farm size, water delivery costs, etc.) are identical or similar across regions. However, existing spatial variation within economies makes that assumption of little use for simulation of real-world policy interventions [191].

Two extensive review papers on CGE and water have been published recently. They provide insight into methodological issues associated with the modeling approaches. The first paper [192] focuses on the implications of using different assumptions regarding water as an implicit or explicit factor of production. They further distinguish within models in which water is assumed an explicit factor of production between assumptions of a high degree of substitution between water and other primary factors, as opposed to low degree of substitution. The paper also differentiates between

regional and global models, including international trade. The review identifies several gaps that need to be considered in the future, such as inclusion of non-consumptive uses, and heterogeneity among the water-using agents or regions. The second review [193] focuses on how the water-energy-food nexus (WEFN) is dealt with in CGE models, discussing their design, importance, and possible ways of improvement. A growing literature is seen lately with focus on WEFN and understanding the modeling structure would be critical for evaluating their results. The paper argues that most CGEs in the literature face difficulties representing the competing water uses across sectors, especially in the energy sector. Thus, addressing the issue of precise distinction across competing water use sectors as a necessary objective in future research.

While policy interventions at the regional (micro) levels could lead to desirable results, local considerations also may lead to a sub-optimal outcome, from a social point of view. This point is demonstrated in recent findings from works on economy-wide considerations and linkages in Morocco, South Africa, Turkey, and Mexico [194–198], respectively, which were summarized and synthesized in [191]. It was found that reforms in sectors other than agriculture have major impacts on rural households' income, and that water reforms that are designed without taking into account reforms outside the major consumer of available water—irrigated agriculture—may lower overall productivity of irrigation water and impede a negative impact on the other sectors competing for the limited resource.

Additional interesting and useful CGE works include studies examining investments in the water sector to address possible investments in reservoirs or water transfers, aimed to relieve regional water scarcity [199], and national investment in extending the water portfolio by considering wastewater treatment facilities and desalination plants [200]. Several studies introduced consideration of new irrigation technologies and irrigation management practices, and their likely effect on the water-using sectors [201,202]). A large body of work focusing on international trade impacts on water-sector performance is the reflection of globalization impacts on national economies [203,204]. It should be mentioned that the analysis by [202,204] is at the global level, rather than at regional or a country levels.

Another brand of studies introduces the use of CGE models for national or sectoral planning by government agencies (Australia [205]; New Zealand: [206]). And finally, one study of many that demonstrated the usefulness of the CGE framework in analyzing economy-wide and sectoral impacts of climate change is [207], which was able to break down economic consequences of climate change by sector and region.

And finally, [208] provides an extensive review of challenges when incorporating water into a CGE model operating at global scale. Challenges are due to absence of standardized data, the sheer overlap caused by intersecting river basins with countries (in the case of international water), and difficulties to model demand for and supply of water of different jurisdictions. Simplifications to the modeling has been introduced to face such challenges. The paper refers to the three most used simplifications in the literature: (a) tackling global questions in a national level model; (b) collapsing irrigated and rainfed crop production into a single sector; and (c) removing river basin boundaries within a country. The paper compares the impact of such simplifications on the ability to predict impacts of future irrigation water scarcity on land use, crop output, international trade, and regional welfare, relative to the full-scale model. Findings suggest that it may be sufficient to ignore the subnational hydrological boundaries in global economic CGE analysis.

7. Linking all the above into one framework

Water suppliers and the water-using sectors face major challenges due to external (climate change, globalization) and internal (population growth, failing institutions) interruptions in water availability and quality. This review has identified several ways in which scholars in the field handled such challenges at local, regional and, sometimes, global levels.

One of the lessons from reading the literature calls for work that integrates the entire set of water types—surface water, groundwater, wastewater, and desalinated water. Some of the works cited demonstrate the conjunctive use of some of these water types, but not all of them. Having models with use, investment, and tradeoff among those types of water demonstrates the usefulness of such integration, especially when the water supply sources are sensitive to climate change [6]. An important consideration is the role of technological innovations in securing/conserving water

resources and regulating its quality. Many studies investigated the feasibility (physical and economic) of water quality-enhancing and quantity-conserving technologies. But such studies are incomplete in that they cannot guarantee the adoption and use of such technologies so that the water sector can benefit from their performance.

A recent surge in scientific work identified international water as a critical issue in sustainability of water in internationally shared river basins. Given that more than two thirds of the world land area contains international river basins [80], sharing water in a sustainable way becomes critical in many parts of the world. As a common pool and strategic resource at the local and international levels the use of allocation methods that account for behavioral and experimental economics and game theory to deal with strategic issues, are quite important and needed.

It is quite intriguing to think about the possibility of having interactions between the results obtained by tools, such as experimental economics, game theory, valuation, and institutional innovation and modeling frameworks that can demonstrate their use in a basin, a region, and globally. Water resources are of different natures, being valued differently by different users, and being considered by different users a strategic resource to different extents. Therefore, a framework that includes various analytical approaches would contribute more than a single approach to addressing challenges faced by water users.

8. Unaddressed issues and agenda for future research

It is only natural that this review cannot address the entire spectrum of new water issues and how economic and modeling approaches may help. An attempt to cover all water issues and all possible approaches may not be achieved. The listed issues and the discussion below are a good assessment of what is still left to be addressed.

In terms of the types of water and water uses, this review does not cover the area of treated wastewater and desalinated water, although they were included in reviews of works that deal with multi-sectoral and intertemporal considerations. These two sources of water were almost taboo in the early 2000s, but as natural water (precipitation) becomes more unreliable and scarcer, the value of conjunctive usage of surface water, groundwater, treated wastewater, and desalinated water increases dramatically. A recent example is the diversion of the Israeli National Carrier Project from using only snowmelt water, originally stored in the Sea of Galilee and conveyed to Southern Israel, to become the destination of desalinated sea water from seawater treatment facilities along the Mediterranean. This revolutionary concept will allow the water system in Israel not only to sustain the economic activity of the state of Israel, but also to support thirsty Jordan with additional drinking and irrigation water in a long-term agreement (https://www.timesofisrael.com/israel-to-be-1st-in-world-to-pipe-desalinated-water-into-a-natural-lake-the-galilee/).

This brings us to another aspect—use of technology to increase regional cooperation in water-scarce regions. The situation described in the previous paragraph—involving a technology in sustainability of water resources, and increasing economics and political stability of a water-scarce region—is only one example of the possible use of a technology to enhance cooperation in managing joint water resources. Another technique is cloud seeding, which can be relevant to many international river basins. By setting rules for allocation of the cost of cloud seeding among the riparian states, including compensation for cost of cloud seeding if rain falls in the parts of the basin belonging to a riparian state that did not invest in the seeding.

Moving along the untouched issues brings us to a very acute policy question of dam removal. Recent work on the economics of dam removal has brought to light many controversial opinions that are seen in publications such as [209], in which cited works identified severe flaws in the consideration of the benefits of dam removals and fish passage.

A different perspective on dams is seen in [210], where the state of knowledge about dam values and impacts is reviewed. The paper argues that evidence suggests that investment in additional surface water storage may not always be the best solution for addressing water scarcity. The author suggests that perhaps due to existing heterogeneity in the sector that affects the performance of dams, policymakers do not appear to make much use of economic tools for decision-making of dam operations. This calls for advancing economic analysis of dam evaluation to avoid costly mistakes related to construction and removal of such infrastructures in the future.

And finally, many works deal with groundwater, and even a conjunctive management of groundwater and surface water. Relatively few studies exist that evaluate groundwater policies to prevent negative externalities from groundwater extraction over time. Several directions have been proposed in previous sections of this review. However, given the relatively few works involving economic considerations of GW-led land subsidence [47,211], more needs to be done on the economics of GW-led land subsidence. GW-led water quality crises are occurring in several parts of the world [212,213] and need to be reflected in more economic analyses. Another groundwater-related important direction of work is the regional analysis of managed aquifer recharge (MAR) and how MAR can ease long-term regional water scarcity. Research on MAR combines all components that have been recommended in this review—inclusion of different types of water and allowing regional cooperation and multisector considerations.

Funding: This research received no external funding.

Data Availability Statement: While links to the majority of the referenced publications are provided in the reference list, readers who may face a problem downloading any of the references are invited to approach the author for help.

Acknowledgments: The work leading to this paper took place in winter 2021, when the author was on Sabbatical leave at Columbia University Water Center.

Conflicts of Interest: The author declares no conflict of interest.

References

- 1. Shiklomanov, Igor, A. **1993**. World freshwater resources. In: Peter H. Gleick (Ed.) *Water in Crisis: A guide to the World Fresh Water Resources*. New York: Oxford University Press.
- 2. Shiklomanov, Igor, A. **1996**. *Assessment of Water Resources and Water Availability in the World*. St. Petersburg: State Hydrological Institute.
- 3. Shiklomanov, Igor, A. **1999**. World Water Resources: An Appraisal for the 21st Century. IHP Report. Paris: UNESCO.
- 4. Shiklomanov, Igor, A. **2000**. Appraisal and assessment of world water resources. *Water International*, 25(1):11-32.
- 5. Clarke, Robin, and Janet King. 2004. The Atlas of Water. London: Earthscan.
- 6. Dinar, A., and Y. Tsur. **2021**. *The Economics of Water Resources: A Comprehensive Approach*. Cambridge, UK: Cambridge University Press.
- 7. Boretti, A., and L. Rosa. **2019**. Reassessing the projections of the World Water Development Report. *npj Clean Water* **2**, 15. https://doi.org/10.1038/s41545-019-0039-9.
- 8. Konapala, G., Mishra, A.K., Wada, Y. et al. **2020**. Climate change will affect global water availability through compounding changes in seasonal precipitation and evaporation. Natural Communication 11, 3044. https://doi.org/10.1038/s41467-020-16757-w.
- 9. Dinar, A., A. Tieu, and H. Huynh. **2019**. Water scarcity impacts on global food production. *Global Food Security*, 23:212-226. https://doi.org/10.1016/j.gfs.2019.07.007.
- 10. Damania, R. **2020**. The economics of water scarcity and variability. *Oxford Review of Economic Policy*, 36(1), 24-44. https://doi.org/10.1093/oxrep/grz027.
- 11. Barbier, E. B. 2004. Water and economic growth. Economic Record, 80(248), 1–16.
- 12. Barbier, E. B. **2015**. Water and growth in developing countries, in A. Dinar and K. Schwabe (Eds.), *Handbook of Water Economics*, Cheltenham, Edward Elgar, pp. 500-512.
- 13. Olmstead, S. M. **2010a**. The economics of managing scarce water resources. *Review of Environmental Economics and Policy*, 4(2). https://www.journals.uchicago.edu/doi/full/10.1093/reep/req004.
- 14. Garrick, D. E., M. Hanemann, and C. Hepburn. **2020**. Rethinking the economics of water: An assessment. *Oxford Review of Economic Policy*, 36(1), 1-23. https://doi.org/10.1093/oxrep/grz035.
- 15. Jeuland, M. **2021**. Review of the state of the art in analysis of the economics of water resources infrastructure. *Oxford Research Encyclopedia of Environmental Science*. https://doi.org/10.1093/acrefore/9780199389414.013.665.
- 16. UN-Water, n.d. Water Scarcity https://www.unwater.org/water-facts/scarcity/. Accessed on August 11, 2022.
- 17. Dolan, Flannery, Jonathan Lamontagne, Robert Link, Mohamad Hejazi, Patrick Reed, and Jae Edmonds. **2021**. Evaluating the economic impact of water scarcity in a changing world. *Nature Communications* 12, 1915. https://doi.org/10.1038/s41467-021-22194-0.

- 18. Schwarzenbach, R. P., T. Egli, T. B. Hofstetter, U. von Gunten, and B. Wehrli. **2010**. Global water pollution and human health. *Annual Review of Environment and Resources*, 35:109-136. https://www.annualreviews.org/doi/full/10.1146/annurev-environ-100809-125342.
- 19. UNEP (United Nations Environment Programme). **2016**. A snapshot of the world's water quality: Towards a global assessment. Nairobi, Kenya: United Nations Environment Programme. https://wesr.unep.org/media/docs/assessments/unep_wwqa_report_web.pdf.
- 20. Mateo-Sagasta, J., S. M. Zadeh, H. Turral, and J. Burke. **2017**. *Water Pollution from Agriculture: A Global Review*. Rome and Colombo: FAO and IWMI. https://www.fao.org/3/ca0146en/CA0146EN.pdf.
- 21. Lall, U., L. Josset, and T. Russo. **2020**. A snapshot of the world's groundwater challenges. *Annual Review of Environment and Resources*, 45:171-194. https://doi.org/10.1146/annurev-environ-102017-025800.
- 22. Olmstead, S. M. 2010b. The economics of water quality. Review of Environmental Economics and Policy, 4(1).
- 23. Shortle, J. S., and R. D. Horan. **2008**. The economics of water quality trading. *International Review of Environmental and Resource Economics*, 2008(2): 101–133.
- 24. Quinn, N. W.T., A. Dinar, I. Kan, and V. K. Sridharan. 2022. Decision support tools for water quality management, special issue, Water. https://www.mdpi.com/journal/water/special_issues/decision_support_tools.
- 25. Tsur, Y., and A. Zemel. **2017**. Coping with multiple catastrophic threats. *Environmental and Resource Economics*, 68(1): 175–196.
- 26. Tsur, Y., and A. Zemel. **2021**. Resource management under catastrophic threats. *Annual Review of Resource Economics*, 13:403-425. https://doi.org/10.1146/annurev-resource-100920-020522.
- 27. Pourzand, Farnaz, and Ilan Noy. **2022**. Catastrophic droughts and their economic consequences. *Oxford Research Encyclopedias, Environmental Science*. https://doi.org/10.1093/acrefore/9780199389414.013.689, New York: Oxford.
- 28. Wilhite, D. A. 1993. Drought Assessment, Management and Planning: Theory and Case Studies. Boston: Kluwer Academic Publishers.
- 29. Wilhite, D. A. **2000**. *Drought, a Global Assessment* (Volume 1 and 2). Routledge Hazards and Disaster Series. New York: Routledge.
- 30. Cavallo, E. A., and I. Noy. **2009**. The economics of natural disasters: A survey. *International Review of Environmental and Resource Economics*, 5(1), 63–102.
- 31. Gil, M., A. Garrido, and N. Hernández-Mora. **2013**. Direct and indirect economic impacts of drought in the agri-food sector in the Ebro River basin (Spain). *Natural Hazards Earth Systems Science*, 13:2679–2694. https://doi.org/10.5194/nhess-13-2679-2013.
- 32. Hezron, M., S. Gichere, R. Davis, and R. Hirji. **2006**. Climate Variability and Water Resource Degradation in Kenya: Improving Water Resources Development and Management. World Bank Working Paper No. 69. Washington, DC: World Bank. https://openknowledge.worldbank.org/handle/10986/7414.
- 33. Freire-González, J., C. Decker, and J. W. Hall. **2017**. The economic impacts of droughts: A framework for analysis. *Ecological Economics*, 132:196-204. https://doi.org/10.1016/j.ecolecon.2016.11.005.
- 34. Naumann, G., C. Cammalleri, L. Mentaschi, and L. Feyen. **2021**. Increased economic drought impacts in Europe with anthropogenic warming. *Natural Climate Change*. 11:485–491. https://doi.org/10.1038/s41558-021-01044-3
- 35. Desbureaux, S., and A.-S. Rodella. **2018**. Drought in the city: The economic impact of water scarcity in Latin American metropolitan areas. *World Development*, 114:13-27. https://doi.org/10.1016/j.worlddev.2018.09.026.
- 36. Booker, J. F., A. M. Michelsen, and F. A. Ward. **2005**. Economic impact of alternative policy responses to prolonged and severe drought in the Rio Grande Basin. *Water Resources Research*, 41(2), W02026. https://doi.org/10.1029/2004WR003486.
- 37. Allaire, M. **2022**. Editorial Disaster impacts and adaptation: The economics of flooding. *Water Economics and Policy*, 8(3): 2202001. https://doi.org/10.1142/S2382624 × 22020015.
- 38. Sun, Q., J. Mann, and M. Skidmore. **2022**. The impacts of flooding and business activity and employment: A spatial perspective on small business. *Water Economics and Policy*, 8(3): 2140003. https://doi.org/10.1142/S2382624 × 21400038.
- 39. Tonn, G., and J. Czajkowski. **2022**. Evaluating the risk and complexity of pluvial flood damage in the U.S. *Water Economics and Policy*, 8(3): 2240002. https://doi.org/10.1142/S2382624 × 22400021.
- 40. Ahmed, R., W. Barkat, A. Ahmed, M. Tahir, and A. M. Nasir. **2022**. The impact of flooding on education of children and adolescents: Evidence from Pakistan. *Water Economics and Policy*, 8(3): 2240009. https://doi.org/10.1142/S2382624 × 22400094.
- 41. Allaire, M. **2020**. Disparities in disaster assistance: A comparison of the social benefits of flood insurance and compensation. *Water Economics and Policy*, 6(4): 2050007. https://doi.org/10.1142/S2382624 × 20500071.
- 42. Esteban, Encarna. **2021**. Emerging issues in groundwater sustainability: New challenges. Oxford Research Encyclopedias, Environmental Science. https://doi.org/10.1093/acrefore/9780199389414.013.697, New York: Oxford.

- 43. Gleeson, T., Y. Wada, M. F. P. Bierkens, and L. P. H. van Beek. **2012**. Water balance of global aquifers revealed by groundwater footprint. *Nature*, 488, 197–200.
- 44. Koundouri, P., K. Roseta-Palma, and N. Englezos. **2017**. Out of sight, not out of mind: Developments in economic models of groundwater management. *International Review of Environmental and Resource Economics*, 11, 55–96.
- 45. Vörösmarty, C. J., P. B. McIntyre, M. O. Gessner, D. Dudgeon, A. Prusevich, P. Green, S. Glidden, S. E. Bunn, C. Sullivan, R. Liermann, and P. M. Davies. **2010**. Global threats to human water security and river biodiversity. *Nature*, 467, 555–561.
- 46. Schreiner-McGraw, A. P., and H. Ajami. **2021**. Delayed response of groundwater to multi-year meteorological droughts in the absence of anthropogenic management. *Journal of Hydrology*, 603(Part B): 126917. https://doi.org/10.1016/j.jhydrol.2021.126917.
- 47. Herrera-García, G., P. Ezquerro, R. Tomás, M. Béjar-Pizarro, J. López-Vinielles, M. Rossi, R. M. Mateos, D. Carreón-Freyre, J. Lambert, P. Teatini, E. Cabral-Cano, G. Erkens, D. Galloway, W.-C. Hung, N. Kakar, M. Sneed, L. Tosi, H. Wang, and S. Ye. **2021**. Mapping the global threat of land subsidence. *Science*, 371(6524):34-36. https://www.science.org/doi/10.1126/science.abb8549.
- 48. Esteban, E., and J. Albiac. **2011**. Groundwater and ecosystems damages: Questioning the Gisser-Sanchez effect. *Ecological Economics*, 70, 2062–2069.
- 49. Esteban, E., and A. Dinar. **2015**. The role of groundwater-dependent ecosystems in groundwater management. *Natural Resource Modeling*, 29(1):98-129, 2016. https://doi.org/10.1111/nrm.12082.
- 50. Esteban, E., and A. Dinar. **2013a**. Cooperative management of groundwater resources in the presence of environmental externalities. *Environmental and Resource Economics*, 54(3), 443–469.
- 51. Molle, F., and A. Closas. **2019**. Co-management of groundwater: A review. *WIREs Water*, 7(1), e1394. https://doi.org/10.1002/wat2.1394.
- 52. Esteban, E., and A. Dinar. **2013b**. Modeling sustainable groundwater management: Packaging and sequencing of policy interventions. *Journal of Environmental Management*, 119, 93–102.
- 53. Pereau, J. C., A. Pryet, and T. Rambonilaza. **2019**. Optimal versus viability in groundwater management with environmental flows. *Ecological Economics*, *161*, 109–120. https://doi.org/10.1016/j.ecolecon.2019.03.018.
- 54. Pongkijvorasin, S., K. Burnett, and C. Wada. **2018**. Joint management of an interconnected coastal aquifer and invasive tree. *Ecological Economics*, 146, 125–135. https://doi.org/10.1016/j.ecolecon.2017.10.011.
- 55. Reznik, A., A. Dinar, S. Bresney, L. Forni, B. Joyce, S. Wallander, D. Bigelow, and I. Kan. **2022**. Institutions and the economic efficiency of managed aquifer recharge as a mitigation strategy against drought impacts on irrigated agriculture in California. *Water Resources Research*, 58(6). https://agupubs.onlinelibrary.wiley.com/doi/full/10.1029/2021WR031261.
- 56. Reznik, A., A. Dinar, and F. Hernandez-Sancho. **2019**. Treated wastewater reuse: An efficient and sustainable solution for water resource scarcity. *Environmental and Resource Economics*, 74(4):1647–1685. https://doi.org/10.1007/s10640-019-00383-2.
- 57. Reznik, A., and A. Dinar. **2021**. Local conditions and the economic feasibility of urban wastewater recycling in irrigated agriculture: Lessons from a stochastic regional analysis in California. *Applied Economic Perspectives and Policy*, 44(4): 2115-2130. https://doi.org/10.1002/aepp.13198.
- 58. Reznik, A., Y. Jiang, and A. Dinar. **2020**. The impacts of climate change on wastewater treatment costs: Evidence from the wastewater sector in China. *Water*, 12(11):3272, https://doi.org/10.3390/w12113272.
- 59. Younos, T. **2005**. The economics of desalination. *Journal of Contemporary Water Research and Education*, 132: 39-45. https://opensiuc.lib.siu.edu/cgi/viewcontent.cgi?article=1050&context=jcwre.
- 60. Sewilam, H., and P. Nasr. **2017**. Desalinated water for food production in the Arab Region. Chapter 4 In: Amer, K. M., Z. Adeel, B. Boer, W. Saleh (Eds.), *The Water, Energy, and Food Security Nexus in the Arab World*. pp. 59-81. https://link.springer.com/chapter/10.1007/978-3-319-48408-2_4.
- 61. Kaner, A., E. Tripler, E. Hadas, and A. Ben-Gal. **2017**. Feasibility of desalination as an alternative to irrigation with water high in salts. *Desalination*, 416: 122-128. https://doi.org/10.1016/j.desal.2017.05.002.
- 62. Inman, D., and P. Jeffrey. **2006**. A review of residential water conservation tool performance and influences on implementation effectiveness. *Urban Water Journal*, 3(3), 127-143. https://doi.org/10.1080/15730620600961288.
- 63. Perry, C., and P. Steduto. **2017**. Does Improved Irrigation Technology Save Water? A Review of the Evidence. FAO Discussion Paper, Cairo: FAO. https://www.fao.org/3/I7090EN/i7090en.pdf.
- 64. Pérez-Blanco, C. D., A. Hrast-Essenfelder, and C. Perry. **2020**. Irrigation technology and water conservation: A review of the theory and evidence. *Review of Environmental Economics and Policy*, 14(2). https://doi.org/10.1093/reep/reaa004.
- 65. Ward, F. A., and M. Pulido-Velazquez. **2008**. Water conservation in irrigation can increase water use. *PNAS*, 105(47):18215-18220. https://doi.org/10.1073/pnas.0805554105.
- 66. Grafton, R. Q., J. Williams, C. J. Perry, F. Molle, C. Ringler, P. Steduto, B. Udall, S. A. Wheeler, Y. Wang, D. Garrick, and R. G. Allen. **2018**. The paradox of irrigation efficiency. *Science*, 361(6404): 748-750. https://www.science.org/doi/10.1126/science.aat9314.

- 67. Dinar, A. and D. Zilberman (Eds.), **1991**. *The economics and management of water and drainage in agriculture.* Kluwer Academic Publishers, Boston, MA.
- 68. Siderius. C., H. Biemans, D. Conway, W. Immerzeel, J. Jaegermeyr, B. Ahmad, and P. Hellegers. **2021**. Financial feasibility of water conservation in agriculture. *Earth's Future*, 9(3): e2020EF001726. https://doi.org/10.1029/2020EF001726.
- 69. Vatta, K., R. S. Sidhu, U. Lall, P. S. Birthal, G. Taneja, B. Kaur, N. Devineni, and C. MacAlister. **2018**. Assessing the economic impact of a low-cost water-saving irrigation technology in Indian Punjab: The tensiometer, *Water International*, 43(2): 305-321. https://doi.org/10.1080/02508060.2017.1416443.
- 70. Nauges, C. and D. Whittington. **2017**. Evaluating the performance of alternative municipal water tariff designs: Quantifying the tradeoffs between equity, economic efficiency, and cost recovery. *World Development*, 91: 125-143. https://doi.org/10.1016/j.worlddev.2016.10.014
- 71. Lu, L., D. Deller and M. Hviid. **2019**. Price and behavioural signals to encourage household water conservation: Implications for the UK. *Water Resources Management*, 33: 475–491. (2019). https://doi.org/10.1007/s11269-018-2133-z
- 72. Goffi, A. S., F. Trojan, J. D. de Lima, M. Lizot, and S. S. Thesari. **2018**. Crossmark economic feasibility for selecting wastewater treatment systems. *Water Science and Technology*, 78 (12): 2518–2531. https://doi.org/10.2166/wst.2019.012.
- 73. Mayzelle, M. M., J. H. Viers, J. Medellín-Azuara, and T. Harter. **2015**. Economic feasibility of irrigated agricultural land use buffers to reduce groundwater nitrate in rural drinking water sources. *Water*, 7(1): 12-37. https://doi.org/10.3390/w7010012.
- 74. Schaible, G. D., and M. l. P. Aillery. **2012**. *Water Conservation in Irrigated Agriculture: Trends and Challenges in the Face of Emerging Demands*, EIB-99, U.S. Department of Agriculture, Economic Research Service, Washington DC. https://www.ers.usda.gov/webdocs/publications/44696/30956_eib99.pdf.
- 75. de Witt, M., W. P. de Clercq, F. J. B. Velazquez, F. Altobelli, and A. D. Marta. **2021**. An in-depth evaluation of personal barriers to technology adoption in irrigated agriculture in South Africa. *Outlook on Agriculture*, 50(3). https://doi.org/10.1177/0030727020986941.
- 76. Gui, X., and Z. Gou. **2021**. Regional differences in household water technology adoption: A longitudinal study of Building Sustainability Index-certified dwelling units in New South Wales, Australia. *Journal of Cleaner Production*, 307:127338. https://doi.org/10.1016/j.jclepro.2021.127338.
- 77. Warner, L. A., A. J. Lamm, and Colby Silvert. **2020**. Diffusion of water-saving irrigation innovations in Florida's urban residential landscapes. *Urban Forestry & Urban Greening*, 47: 126540. https://doi.org/10.1016/j.ufug.2019.126540.
- 78. Reints, J., A. Dinar, and D. Crowley. **2020**. Dealing with water scarcity and salinity: Adoption of water efficient technologies and management practices by California Avocado Growers. *Sustainability*, 12(9): 3555. https://doi.org/10.3390/su12093555.
- 79. Mukherjee, M., and K. Schwabe, **2015**. Irrigated Agricultural Adaptation to Water and Climate Variability: The Economic Value of a Water Portfolio. *American Journal of Agricultural Economics*, 97(3):809-832. https://doi.org/10.1093/ajae/aau101.
- 80. McCracken, M., and A. T. Wolf. **2019**. Updating the register of international river basins of the world. *International Journal of Water Resources Development*, 35(5): 732–782. https://doi.org/10.1080/07900627.2019.1572497.
- 81. Dinar, S., and A. Dinar. **2003**. Recent developments in the literature on conflict and cooperation in international shared water. *Natural Resources Journal*, 43(4):1217-1287.
- 82. Dinar, A., and M. Hogarth. **2015**. Game theory and water resources: Critical review of its contributions, progress and remaining challenges. *Foundations & Trends in Microeconomics*, 11(1–2):1–139.
- 83. Dinar, S. **2020**. Emerging issues and challenges in transboundary freshwater: The role of treaties and treaty design. *Oxford Research Encyclopedias, Environmental Science*. https://doi.org/10.1093/acrefore/9780199389414.013.698, New York: Oxford.
- 84. Dinar, A., L. De Stefano, G. Nigatu, and N. Zawahri. **2019**. Why are there so few basin-wide treaties? Economics and politics of coalition formation in multilateral international basins. *Water International*, 2019, 44(4):463-485.
- 85. Zawahri, N., A. Dinar, and G. Nigatu. **2014**. Governing international freshwater resources: An analysis of treaty design. *International Environmental Agreements: Politics, Law and Economics*, 16(2):307–331.
- 86. Drezner, D. **2000**. Bargaining, enforcement, and multilateral economic sanctions. *International Organization*, 54:73–102. https://doi.org/10.1162/002081800551127.
- 87. Verdier, D. **2008**. Multilateralism, bilateralism, and exclusion in the nuclear proliferation regime. *International Organization*, 62(3):439–476.
- 88. Dinar, S., A. Dinar, and P. Kurukulasuriya. **2011**. Scarcity and cooperation along international rivers: An empirical assessment of bilateral treaties. *International Studies Quarterly*, 55(3), 809–833.

- 89. Dinar, A., B. Blankespoor, S. Dinar, and P. Kurukulasuriya. **2010**. Does precipitation and runoff variability affect treaty cooperation between states sharing international bilateral rivers? *Ecological Economics*, 69(12), 2568–2581.
- 90. Dinar, S., D. Katz, L. De Stefano, and B. Blankespoor. **2015**. Climate change, conflict, and cooperation: Global analysis of the effectiveness of international river treaties in addressing water variability. *Political Geography*, **45**, 55–66.
- 91. Dinar, S., D. Katz, L. De Stefano, and B. Blankespoor. **2019**. Do treaties matter? Climate change, water variability, and cooperation along transboundary river basins. *Political Geography*, 69, 162–172.
- 92. Barrett, S. **2003**. Environment and statecraft: The strategy of environmental treaty-making. NY: Oxford University Press.
- 93. Dombrowsky, I. **2010**. The role of intra-water sector issue linkage in the resolution of transboundary water conflicts. *Water International*, 35(2):132–149.
- 94. Pham Do, K. H., and A. Dinar. **2014**. The role of issue linkage in managing non-cooperating basins: The case of the Mekong. *Natural Resource Modeling*, 27(4):492-518.
- 95. Pham Do, K. H., A. Dinar, and D. McKinney. **2012**. Transboundary water management: Can issue linkage help mitigate externalities? *International Game Theory Review*, 14(1):39-59.
- 96. Pham Do, K. H., and A. Dinar. **2017a**. The linkages of energy, water, and land use in Southeast Asia: Challenges and opportunities for the Mekong Region. In: Douglas A., C. Arndt, M. Miller, F. Tarp, and O. Zinaman (Eds.) *The Political Economy of Energy Transitions*, London: Oxford University Press, pp. 511-529.
- 97. Pham Do, K. H., and A. Dinar. **2017b**. Issue linkage: A mechanism for managing conflict, applied to the Mekong Basin. In: Dinar, A. and Y. Tsur (Eds.), *Management of Transboundary Water Resources Under Scarcity: A Multidisciplinary Approach*. Singapore: World Scientific, pp. 91-114.
- 98. Katz, D., and A. Shafran. **2020**. Energizing mid–east water diplomacy: The potential for regional water–energy exchanges. *Water International*, 45(4): 292-310, DOI:10.1080/02508060.2020.1758521.
- 99. Walschot, M., and D. Katz. 2022. Desalination and transboundary water conflict and cooperation: A mixed-method empirical approach. *Water* 14: 1925. https://doi.org/10.3390/w14121925
- 100. Banerjee, S. **2022**. Use of experimental economics in policy design and evaluation: An application to water resources and other environmental domains. *Oxford Research Encyclopedias, Environmental Science*. New York: Oxford. https://doi.org/10.1093/acrefore/9780199389414.013.764.
- 101. Bejarano, H., and J. Shortle. **2015**. Experimental economics and water resources. In A. Dinar and K. Schwabe (Eds.), *Handbook of Water Economics*, Cheltenham: Edward Elgar. pp.263-282. https://doi.org/10.4337/9781782549666.00023.
- 102. Meinzen-Dick, R., M. A. Janssen, S. Kandikuppa, R. Chaturvedi, K. Rao, and S. Theis. **2018**. Playing games to save water: Collective action games for groundwater management in Andhra Pradesh, India. *World Development*, 107: 40–53. https://doi.org/10.1016/j.worlddev.2018.02.006.
- 103. Torres, M. M. J., and F. Carlsson, F. **2018**. Direct and spillover effects of a social information campaign on residential water-savings. *Journal of Environmental Economics and Management*, 92: 222–243. https://doi.org/10.1016/j.jeem.2018.08.005.
- 104. Brent, D. A., C. Lott, M. Taylor, J. Cook, K. Rollins, and S. Stoddard. **2020**. What causes heterogeneous responses to social comparison messages for water conservation? *Environmental and Resource Economics*, 77(3):503–537. https://doi.org/10.1007/s10640-020-00506-0.
- 105. Murphy, J. J., A. Dinar, R. E. Howitt, S. J. Rassenti, and V. L. Smith. **2000**. The design of "smart" water market institutions using laboratory experiments. *Environmental and Resource Economics*, 17(4), 375–394. https://doi.org/10.1023/A:1026598014870.
- 106. Murphy, J. J., A. Dinar, R. Howitt, S. Rassenti, V. Smith, and M. Weinberg. 2009. Incorporating instream flow values into a water market. *Journal of Environmental Management*, 90(2):1089-1096. https://reader.elsevier.com/reader/sd/pii/S0301479708001023?token=59D55877A167A00E11CCDF28FCB1 C5DCBE0877E0688B4AAAD9AE3698DC66F02BB520D32BA60A4A8C0A223BC7FCB38518&originRegion =us-east-1&originCreation=20230103001225.
- 107. Suter, J. F., J. M. Spraggon, and G. L. Poe. **2013**. Thin and lumpy: An experimental investigation of water quality trading. *Water Resources and Economics*, 1:36–60. https://doi.org/10.1016/j.wre.2013.03.004.
- 108. Palm-Forster, L. H., J. F. Suter, and K. D. Messer. 2019. Experimental evidence on policy approaches that link agricultural subsidies to water quality outcomes. *American Journal of Agricultural Economics*, 101(1):109–133. https://doi.org/10.1093/ajae/aay057.
- 109. Tellez-Foster E., A. Rapoport, and A. Dinar. **2017a**. Alternative policies to manage electricity subsidies for groundwater extraction: A field study in Mexico. *Journal of Behavioral Economics for Policy*, **2**(2):55-69.
- 110. Tellez-Foster E., A. Rapoport, and A. Dinar. **2017b**. Groundwater and electricity consumption under alternative subsidies: Evidence from laboratory experiments. *Journal of Behavioral and Experimental Economics*, 68:41–52. https://doi.org/10.1016/j.socec.2017.03.003.

- 111. Tellez-Foster, E., A. Dinar, and A. Rapoport. **2018**. Comparing alternative policies for modification of energy subsidies: The case of groundwater pumping for irrigation. *Journal of Hydrology*, 2018, 565:614-622. https://doi.org/10.1016/j.jhydrol.2018.08.071.
- 112. Cardenas, J. C., L. A. Rodriguez, and N. Johnson. **2011**. Collective action for watershed management: Field experiments in Colombia and Kenya, *Environment and Development Economics*, 16(3): 275–303. https://doi.org/10.1017/S1355770 × 10000392.
- 113. Ert, E., S. Cohen-Amin, and A. Dinar. **2019**. The effect of issue linkage on cooperation in bilateral conflicts: An experimental analysis. *Journal of Behavioral and Experimental Economics*, 9:134-142. https://doi.org/10.1016/j.socec.2019.02.004.
- 114. Dinar, A., J. Albiac, and J. Sanchez-Soriano (Eds.). **2008a**. *Game Theory and Policymaking in Natural Resources and the Environment*. New York: Routledge Explorations in Environmental Economics.
- 115. Dinar, A., and A. Rapoport (Eds.). **2013**. *Analyzing Global Environmental Issues: Theoretical and Experimental Applications and Their Policy Implications*. New York: Routledge.
- 116. Pham Do, Kim Hang. **2022**. Water resource management: Challenges and opportunities with game theory approaches. *Oxford Research Encyclopedias, Environmental Science*. https://doi.org/10.1093/acrefore/9780199389414.013.763, New York: Oxford.
- 117. Dinar, A., S. Farolfi, F. Patrone, and K. Rowntree. **2008b**. To negotiate or to game theorize: Evaluating water allocation mechanisms in the Kat Basin, South Africa. Chapter 5 in Dinar, A., J. Albiac, and J. Sanchez-Soriano (Eds.) *Game Theory and Policymaking in Natural Resources and the Environment*. New York: Routledge Explorations in Environmental Economics. pp. 85-111.
- 118. Li, X., H. Shi, and X. Lin. **2008**. Game theoretic modeling of water allocation regimes applied to the Yellow River Basin in China. Chapter 12 in Dinar, A., J. Albiac, and J. Sanchez-Soriano (Eds.) *Game Theory and Policymaking in Natural Resources and the Environment*. New York: Routledge Explorations in Environmental Economics. pp. 248-265.
- 119. Ambec, S., and L. Ehlers. **2008**. Cooperation and equity in the river-sharing problem. Chapter 6 in Dinar, A., J. Albiac, and J. Sanchez-Soriano (Eds.), *Game Theory and Policymaking in Natural Resources and the Environment*. New York: Routledge Explorations in Environmental Economics. pp. 112-131.
- 120. Ambec, S., A. Dinar, and D. McKinney. **2013**. Water sharing agreements sustainable to reduced flows. *Journal of Environmental Economics and Management*, 66(3):639–655.
- 121. Dinar, A., and G. Nigatu. **2013**. Distributional considerations of international water resources un-der externality: The case of Ethiopia, Sudan and Egypt on the Blue Nile. *Water Resources & Economics*, 2–3:1–16.
- 122. Nigatu, G., and A. Dinar. **2016**. Economic and hydrological impacts of the Grand Ethiopian Re-naissance Dam on the Eastern Nile River Basin. *Environment and Development Economics*, 21(4):532 555.
- 123. Frisvold, G. B., and K. J. Emerick. **2008**. Rural-urban water transfers with applications to the US-Mexico border region. Chapter 7 in Dinar, A., J. Albiac, and J. Sanchez-Soriano (Eds.) *Game Theory and Policymaking in Natural Resources and the Environment*. New York: Routledge Explorations in Environmental Economics. pp. 155-180.
- 124. Fisher, M. F., and Annette T. Huber-Lee. **2008**. WAS-guided cooperation in water management: Coalitions and gains. Chapter 9 in Dinar, A., J. Albiac, and J. Sanchez-Soriano (Eds.) *Game Theory and Policymaking in Natural Resources and the Environment*. New York: Routledge Explorations in Environmental Economics. pp. 181-208.
- 125. Llioret, A. **2013**. Informal Agreements in transboundary water resources. Chapter 10 in Dinar, A. and A. Rapoport (Eds.) *Analyzing Global Environmental Issues: Theoretical and Experimental Applications and their Policy Implications*. New York: Routledge. pp: 170-185.
- 126. Zhu, X., H. Houba, and Kim Hang Pham Do. 2013. Efficient use of the Mekong River Basin: A joint management approach. Chapter 11 in Dinar, A. and A. Rapoport (Eds.) *Analyzing Global Environmental Issues: Theoretical and Experimental Applications and their Policy Implications*. New York: Routledge. pp: 186-202.
- 127. Ansink, E., M. Gengenbach, and H. P. Weikard. 2017. River coalitions and water trade. *Oxford Economic Papers*, 69(2), 453–469.
- 128. Pham Do Kim Hang. 2022. Water Resource Management: Challenges and Opportunities with Game Theory Approaches. Oxford Research Encyclopedia.
- 129. https://doi.org/10.1093/acrefore/9780199389414.013.763.
- 130. Madani, K., and A. Dinar. **2012a**. Cooperative institutions for sustainable common pool resource management: Application to groundwater. *Water Resource Research*, 48(9). https://doi.org/10.1029/2011WR010849.
- 131. Madani, K., and A. Dinar. **2012b**. Non-cooperative institutions for sustainable common pool resource management: Application to groundwater. *Ecological Economics*, 74:34–45.
- 132. Raquel, S., S. Ferenc, C. Emery, Jr., and R. Abraham. **2007**. Application of game theory for a groundwater conflict in Mexico. *Journal of Environmental Management*, 84(4):560–571.

- 133. Jiang, K., R. Merrill, D. You, P. Pan, and Z. Li. **2019**. Optimal control for transboundary pollution under ecological compensation: A stochastic differential game approach. *Journal of Cleaner Production*, 241:118391. https://doi.org/10.1016/j.jclepro.2019.118391.
- 134. Xiao, L., J. Liu, and J. Ge. **2021**. Dynamic game in agriculture and industry cross-sectoral water pollution governance in developing countries. *Agricultural Water Management*, 243: 106417. https://doi.org/10.1016/j.agwat.2020.106417.
- 135. Sears, L., D. Lim, and C.-Y. L. Lawell. **2019**. Spatial groundwater management: A dynamic game framework and application to California. *Water Economics and Policy*, 5(1): 1850019, https://doi.org/10.1142/S2382624 × 18500194.
- 136. Hemati, H., and A. Abrishamchi. **2021**. Water allocation using game theory under climate change impact (case study: Zarinehrood). *Journal of Water and Climate Change*, 12(3):759–771.
- 137. Rightnar, J., and A. Dinar. **2020**. The welfare implications of bankruptcy allocation of the Colorado River water: The case of the Salton Sea Region. *Water Resources Management*, 34:2353–2370. https://doi.org/10.1007/s11269-020-02552-1.
- 138. Moretti, S., F. Patrone, A. Dinar, and S. Abdel-Dayem, **2016**. Sharing the Costs of Complex Water Projects: Application to the West Delta Water Conservation and Irrigation Rehabilitation Project, Egypt. *Games*, 7(3): 18. https://doi.org/10.3390/g7030018.
- 139. Meinzen-Dick, R. **2007**. Beyond panaceas in water institutions. PNAS, 104(39): 15200-15205. https://doi.org/10.1073/pnas.0702296104.
- 140. Dinar, A., K. Kemper, W. Blomquist, and P. Kurukulasuriya. **2007**. Whitewater: Process and performance of decentralization reform of river basin water resource management. *Journal of Policy Modeling*, 29(6):851-867. https://doi.org/10.1016/j.jpolmod.2007.06.013.
- 141. Dinar, A., J. Ortiz-Correa, S. Farolfi, J. Mutondo. **2016**. Quantifying the process and performance of river basin water management decentralization in Sub-Saharan Africa. *Journal of African Economies*, 25(2): 267-299. https://doi.org/10.1093/jae/ejv023.
- 142. Maria-Saleth, R., and A. Dinar. **2004**. *The Institutional Economics of Water: A Cross-Country Analysis of Institutions and Performance*. Cheltenham: Edward Elgar. https://openknowledge.worldbank.org/handle/10986/14884.
- 143. Araral, E., and D. J. Yu. **2013**. Comparative water law, policies, and administration in Asia: Evidence from 17 countries. *Water Resources Research*, 49(9): 5307-5316. https://doi.org/10.1002/wrcr.20414.
- 144. Bromley, D. W., and G. Anderson. **2018**. Does water governance matter? *Water Economics and Policy*, **4**(3): 1750002. https://doi.org/10.1142/S2382624 × 17500023.
- 145. Dinar, A., and R. Maria Saleth. **2005**. Can water institution be cured? A water institutions health index. *Water Science and Technology: Water Supply Journal* 5(6):17-40. https://doi.org/10.2166/ws.2005.0047.
- 146. Maria-Saleth, R., and A. Dinar. **2008**. Linkages within institutional structure: An empirical analysis of water institutions. *Journal of Institutional Economics*, 4(3):375-401. https://doi.org/10.1017/S1744137408001136.
- 147. Bandaragoda, D. J. **2000**. A Framework for Institutional Analysis for Water Resources Management in a River Basin Context. IWMI Working Paper 5. https://www.iwmi.cgiar.org/Publications/Working_Papers/working/WOR5.pdf.
- 148. Maria-Saleth, R. and A. Dinar. 1999a. Water challenge and institutional response. (A Cross Country Perspective), Policy Research Working Paper 2045. Washington D.C.: The World Bank.
- 149. Maria-Saleth, R. and A. Dinar. **1999b**. Evaluating water institutions and water sector performance. World Bank Technical Paper No. 447. Washington D.C.: The World Bank.
- 150. Berbel, J., N. M. Montilla-López, G. Giannoccaro. 2020. Institutions and economics of water scarcity and droughts (Special Issue). *Water*, 12(S11): 3248. https://doi.org/10.3390/w12113248.
- 151. Chopra, A., and P. Ramachandran. **2021**. Understanding water institutions and their impact on the performance of the water sector in India. *Water Policy*, 23(2): 466–486. https://doi.org/10.2166/wp.2021.207.
- 152. Adamowicz, W., L. Calderon-Etter, A. Entem, E. P. Fenichel, J. S. Hall, P. Lloyd-Smith, and R. F. Stallard. **2019**. Assessing ecological infrastructure investments. *Proceedings of the National Academy of Sciences*, *116*(12), 5254–5261. https://doi.org/10.1073/pnas.1802883116.
- 153. Carson, R. T. **2012**. Contingent valuation: A practical alternative when prices aren't available. *Journal of Economic Perspectives*, 26 (4):27-42. http://dx.doi.org/10.1257/jep.26.4.27.
- 154. Freeman, M. III, J. Herriges, and K. C. Kling. **2014**. *The Measurement of Environmental and Resource Values* (3rd ed.). New York: Resource for the Future Press.
- 155. Carson, R. T., R. C. Mitchell, M. Hanemann, R. J. Kopp, S. Presser, and P. A. Ruud. **2003**. Contingent valuation and lost passive use: Damages from the Exxon Valdez oil spill. *Environmental and Resource Economics* 25:257–286. https://econweb.ucsd.edu/~rcarson/papers/ExxonERE.pdf.
- 156. Kling, C., D. Phaneuf, and J. Zhao. **2012**. From Exxon to BP: Has some number become better than no number? *Journal of Economic Perspectives*, 26(4), 3–26.

- 157. Bishop, R. C., K. J. Boyle, R. T. Carson, D. Chapman, W. M. Hanemann, B. Kanninen, and N. Scherer, N. 2017. Putting a value on injuries to natural assets: The BP oil spill. *Science*, 356(6335), 253–254. https://www.science.org/doi/10.1126/science.aam8124
- 158. Adamowicz, Wic, and Diane Dupont. **2019**. Challenges to environmental valuation of water in light of global change. *Oxford Research Encyclopedias, Environmental Science*. https://doi.org/10.1093/acrefore/9780199389414.013.610, New York: Oxford.
- 159. Zhang, J., W. Adamowicz, D. Dupont, and A. Krupnick. **2013**. Assessing the extent of altruism in the valuation of community drinking water quality improvements. *Water Resources Research*, 49(10), 6286–6297. https://doi.org/10.1002/wrcr.20447.
- 160. Jin, J., Z. Wang, and S. Ran. **2005**. Comparison of contingent valuation and choice experiment in solid waste management programs in Macao. *Ecological Economics*, 57(3):430-441. https://doi.org/10.1016/j.ecolecon.2005.04.020.
- 161. Johnston, R. J., K. J. Boyle, W. Adamowicz, J. Bennett, R. Brouwer, T.A. Cameron, and C. A. Vossler. 2017. Contemporary guidance for stated preference studies. *Journal of the Association of Environmental and Resource Economists*, 4(2), 319-405. https://www.journals.uchicago.edu/doi/epdf/10.1086/691697.
- 162. Bishop, R., and K. Boyle. **2019**. Reliability and validity in nonmarket valuation. *Environmental and Resource Economics*, 72(2), 559–582. https://link.springer.com/article/10.1007/s10640-017-0215-7.
- 163. Lloyd-Smith, P., E. Zawojska, and W. Adamowicz. **2020**. Moving beyond the contingent valuation versus choice experiment debate: Presentation effects in stated preference. *Land Economics*, 96(1): 1-24; DOI: https://doi.org/10.3368/le.96.1.1.
- 164. Tourkolias, C., T. Skiada, S. Mirasgedis, and D. Diakoulaki. **2014**. Application of the travel cost method for the valuation of the Poseidon temple in Sounio, Greece. *Journal of Cultural Heritage*, 16(4): 567-574. https://doi.org/10.1016/j.culher.2014.09.011.
- 165. Czajkowski, M., M. Giergiczny, J. Kronenberg, and J. Englin. **2019**. The individual travel cost method with consumer-specific values of travel time savings. *Environmental and Resource Economics*, 74: 961–984. https://doi.org/10.1007/s10640-019-00355-6.
- 166. Remoundou, K., P. Diaz-Simal, P. Koundouri, and B. Rulleauf. **2015**. Valuing climate change mitigation: A choice experiment on a coastal and marine ecosystem. *Ecosystem Services*, 11: 87-94. https://doi.org/10.1016/j.ecoser.2014.11.003.
- 167. Birol, E., K. Karousakis, and P. Koundouri. **2006**. Using a choice experiment to account for preference heterogeneity in wetland attributes: The case of Cheimaditida wetland in Greece. *Ecological Economics*, 60(1): 145-156. https://doi.org/10.1016/j.ecolecon.2006.06.002.
- 168. Das S., H. Fuchs, R. Philip, and P. Rao. **2023**. A review of water valuation metrics: Supporting sustainable water use in manufacturing. *Water Resources and Industry*, **29**, 100199. https://doi.org/10.1016/j.wri.2022.100199.
- 169. Pulido-Velazquez, M., and A. Tilmant. **2022**. Hydroeconomics. *Oxford Encyclopedia of Environmental Sciences*. https://doi.org/10.1093/acrefore/9780199389414.013.686.
- 170. Harou, J. J., M. Pulido-Velazquez, D. E. Rosenberg, J. Medellín-Azuara, J. R. Lund, and R. E. Howitt. **2009**. Hydro-economic models: Concepts, design, applications, and future prospects. *Journal of Hydrology*, 375(3–4): 627-643. https://doi.org/10.1016/j.jhydrol.2009.06.037.
- 171. Bekchanov, M., A. Sood, A. Pinto, and M. Jeuland. **2017**. Systematic review of water-economy modeling applications. *Journal of Water Resources Planning and Management*, 143(8), 04017037. https://doi.org/10.1061/(ASCE)WR.1943-5452.0000793.
- 172. Brouwer, R., and M. Hofkes. **2008**. Integrated hydro-economic modelling: Approaches, key issues and future research directions. *Ecological Economics*, 66(1), 16–22. https://doi.org/10.1016/j.ecolecon.2008.02.009.
- 173. Eamen, L., R. Brouwer, and S. Razavi. **2022**. Comparing the applicability of hydro-economic modelling approaches for large-scale decision-making in multi-sectoral and multi-regional river basins. *Environmental Modelling & Software*, 152, 105385.
- 174. Dalcin, A. P., and M. G. Fernandes. **2020**. Integrating water management instruments to reconcile a hydroeconomic water allocation strategy with other water preferences. *Water Resources Research*, 56(5): e2019WR025558. https://doi.org/10.1029/2019WR025558.
- 175. Hossen, M. A., J. Connor, and F. Ahammed. **2021**. Review of hydro-economic models (HEMs) which focus on transboundary river water sharing disputes. *Water Policy*, 23(6): 1359–1374. https://doi.org/10.2166/wp.2021.114.
- 176. Arjoon, D., A. Tilmant, and M. Herrmann. **2016**. Sharing water and benefits in transboundary river basins. *Hydrology and Earth System Sciences*, 20(6), 2135–2150. https://doi.org/10.5194/hess-20-2135-2016.
- 177. Kahil, T., S. Parkinson, Y. Satoh, P. Greve, P. Burek, T. I. E. Veldkamp, R. Burtscher, E. Byers, N. Djilali, G. Fischer, V. Krey, S. Langan, K. Riahi, S. Tramberend, and Y. Wada. 2018. A continental-scale hydroeconomic model for integrating water-energy-land nexus solutions. *Water Resources* 54(10): 7511-7533. https://doi.org/10.1029/2017WR022478.

- 178. Kuwayama, Y., and S. M. Olmstead. **2020**. Hydroeconomic modeling of resource recovery from wastewater: Implications for water quality and quantity management. *Journal of Environmental Quality*, 49(3): 593-602. https://doi.org/10.1002/jeq2.20050.
- 179. Baccour, S., J. Albiac, T. Kahil, E. Esteban, D. Crespo, and A. Dinar. **2021**. Hydroeconomic modeling for assessing water scarcity and agricultural pollution abatement policies in the Ebro River Basin, Spain. *Journal of Cleaner Production*, 327. https://doi.org/10.1016/j.jclepro.2021.129459.
- 180. Berthet, A., A. Vincent, and P. Fleury. **2021**. Water quality issues and agriculture: An international review of innovative policy schemes. *Land Use Policy*, 109. Scopus. https://doi.org/10.1016/j.landusepol.2021.105654.
- 181. Crespo, D., J. Albiac, A. Dinar, E. Esteban, and T. Kahil. **2022**. Integrating ecosystem benefits for sustainable water allocation in hydroeconomic modeling. *Plos One*. https://doi.org/10.1371/journal.pone.0267439
- 182. Crespo, D., J. Albiac, T. Kahil, E. Esteban, and S. Baccour. **2019**. Tradeoffs between water uses and environmental flows: A hydroeconomic analysis in the Ebro Basin. *Water Resources Management*, 33, pages2301–2317. https://doi.org/10.1007/s11269-019-02254-3.
- 183. Levers, L. R., T. H. Skaggs, and K. A. Schwabe. 2019. Buying water for the environment: A hydro-economic analysis of Salton Sea inflows. *Agricultural Water Management*, 213, 554–567. https://doi.org/10.1016/j.agwat.2018.10.041.
- 184. Momblanch, A., J. D. Connor, N. D. Crossman, J. Paredes-Arquiola, and J. Andreu. **2016**. Using ecosystem services to represent the environment in hydro-economic models. *Journal of Hydrology*, 538, 293–303. https://doi.org/10.1016/j.jhydrol.2016.04.019.
- 185. MacEwan, D., M. Cayar, A. Taghavi, D. Mitchell, S. Hatchett, and R. Howitt. **2017**. Hydroeconomic modeling of sustainable groundwater management. *Water Resources Research*. https://doi.org/10.1002/2016WR019639.
- 186. Pulido-Velazquez, M., G. F. Marques, J. J. Harou, and J. R. **2016**. Hydroeconomic models as decision support tools for conjunctive management of surface and groundwater. In: Jakeman, A.J., Barreteau, O., Hunt, R.J., Rinaudo, JD., Ross, A. (Eds.) *Integrated Groundwater Management*. Springer, Cham. https://doi.org/10.1007/978-3-319-23576-9_27.
- 187. Kahil, M. T., Ward, F. A., Albiac, J., Eggleston, J., Sanz, D. **2016**. Hydro-economic modeling with aquiferriver interactions to guide sustainable basin management. Journal of Hydrology, 539, 510–524. https://doi.org/10.1016/j.jhydrol.2016.05.057.
- 188. Gürlük, S., and F. A. Ward, **2009**. Integrated Basin Management: Water and Food Policy Options for Turkey. *Ecological Economics*, 68:2666–2678.
- 189. Esteve, P., C. Varela-Ortega, I. Blanco-Gutiérrez, and T. E. Downing. **2015**. A hydro-economic model for the assessment of climate change impacts and adaptation in irrigated agriculture. *Ecological Economics*, 120, 49–58. https://doi.org/10.1016/J.ECOLECON.2015.09.017
- 190. Expósito, A., F. Beier, and J. Berbel. **2020**. Hydro-economic modelling for water-policy assessment under climate change at a river basin scale: A review. *Water*, 12(6), 1559. https://doi.org/10.3390/w12061559.
- 191. Kahil, M. T., A. Dinar, and J. Albiac. **2015**. Modeling water scarcity and droughts for policy adaptation to climate change in arid and semiarid regions. *Journal of Hydrology*, 522, 95–109. https://doi.org/10.1016/j.jhydrol.2014.12.042.
- 192. Dinar, A. **2014**. Water and economy-wide policy interventions. *Foundations and Trends in Microeconomics*, 10(2):1-84. https://escholarship.org/uc/item/0f47p42h.
- 193. Calzadilla, A., K. Rehdanz, R. Roson, M. Sartori, and R. S. J. Tol. **2016**. Review of CGE models of water issues. In: Bryant, T. (Ed.), *The WSPC Reference on Natural Resources and Environmental Policy in the Era of Global Change*, pp. 101-123. https://doi.org/10.1142cadzi/9789813208179_0004.
- 194. Bardazzi, E., and F. Bosello. **2021**. Critical reflections on water-energy-food nexus in computable general equilibrium models: A systematic literature review. *Environmental Modelling & Software*, 145, 105201. https://doi.org/10.1016/j.envsoft.2021.105201.
- 195. Roe, T. L., A. Dinar, Y. Tsur, and X. Diao. **2005**. Feedback links between economy-wide and farm-level policies: With application to irrigation water management in Morocco. *Journal of Policy Modeling* 27(8):905-928. https://doi.org/10.1016/j.jpolmod.2005.05.007.
- 196. Cakmak, E. H., D. Hasan, S. Saracoglu, X. Diao, T. Roe, and Y. Tsur. **2008**. Macro-micro Feedback Links of Irrigation Water Management in Turkey, World Bank Policy Research Working Paper 4781. https://openknowledge.worldbank.org/handle/10986/6350.
- 197. Diao, X., A. Dinar, T. Roe, and Y. Tsur. **2008**. A general equilibrium analysis of conjunctive ground and surface water use with an application to Morocco. *Agricultural Economics* 38:117-135. https://doi.org/10.1111/j.1574-0862.2008.00287.x.
- 198. Yunez-Naude, A., and L. G. Rojas Castro. **2008**. Perspectivas de la agricultura ante reducciones en la disponibilidad de agua para riego: Un enfoque equilibrio general. El Agua en México: Consecuencias de las Políticas de Intervención en el Sector. El Trimestre Económico Lecturas 100:183-211. Fondo del Cultura Económica, México.

- 199. Hassan, R., and J. Thurlow. **2011**. Macro-micro feedback links of water management in South Africa: CGE analyses of selected policy regimes. *Agricultural Economics* 42(2):235-247, 2011. https://doi.org/10.1111/j.1574-0862.2010.00511.x.
- 200. Goodman, D. J. **2000**. More reservoirs or transfers? A computable general equilibrium analysis of projected water shortages in the Arkansas River basin. *Journal of Agricultural and Resource Economics* **25**(2):698-713. https://www.jstor.org/stable/40987085.
- 201. Luckmann, J., H. Grethe, S. McDonald, A. Orlov, and K. Siddig. **2014**. An integrated economic model of multiple types and uses of water. *Water Resources Research*, 50(5): 3875-3892. https://doi.org/10.1002/2013WR014750.
- **202.** Sánchez C., J., and C. Sarasa. **2019**. Uncertainty in irrigation technology: Insights from a CGE approach. *Water* 11, 617. https://doi.org/10.3390/w11030617
- 203. Calzadilla, Alvaro, Katrin Rehdanz, and Richard S. J. Tol. **2011a**. Water scarcity and the impact of improved irrigation management: a computable general equilibrium analysis. *Agricultural Economics* 42(3):305-323. https://doi.org/10.1111/j.1574-0862.2010.00516.x.
- 204. Berrittella, M., K. Rehdanz, R. S. J. Tol, J. Zhang. **2008**. The impact of trade liberalization on water use: A computable general equilibrium analysis. *Journal of Economic Integration* 23(3), 631-655. https://doi.org/10.11130/jei.2008.23.3.631.
- 205. Calzadilla, A., K. Rehndanz, R. S. J. Tol. **2011b**. Trade liberalization and climate change: A computable general equilibrium analysis of the impacts on global agriculture. *Water* 3:526-550. https://doi.org/10.3390/w3020526.
- 206. Wittwer, G. **2012**. *Economic Modeling of Water: The Australian CGE Experience*. Berlin: Springer. https://link.springer.com/book/10.1007/978-94-007-2876-9.
- 207. Kim, J. **2017**. A Computable General Equilibrium (CGE) Approach to Urban Freshwater Planning. Report prepared for the Chief Economist, Unit, Auckland Council, New Zealand. https://environment.govt.nz/assets/Publications/Files/Sub-regional-CGE-model-for-NPSFM-Feasibility-study-final.pdf.
- 208. Ponce, R., R. Parrado, A. Stehr, and F. Bosello. **2016**. Climate change, water scarcity in agriculture and the economy-wide impacts in a CGE framework. *Nota di Lavoro*, 79.2016, Milan, Italy: Fondazione Eni Enrico Mattei. https://www.feem.it/en/publications/climate-change-water-scarcity-in-agriculture-and-the-economy-wide-impacts-in-a-cge-framework/.
- 209. Liu, J., T. Hertel, and F. Taheripour. **2016**. Analyzing future water scarcity in computable general equilibrium models. *Water Economics and Policy*, 2(04), 1650006. https://doi.org/10.1142/S2382624 × 16500065.
- 210. Lewis, L. Y. **2020**. Rethinking hydropower: The economics and politics of privately owned hydropower in the United States. *Oxford Research Encyclopedias, Environmental Science,* Oxford: NY. https://doi.org/10.1093/acrefore/9780199389414.013.691.
- 211. Jeuland, M. **2020**. The economics of dams. *Oxford Review of Economic Policy*, 36(1), 45-68. https://doi.org/10.1093/oxrep/grz028.
- 212. Esteban, E., A. Dinar, E. Calvo Calzada; J. Albiac, J. Calatrava, G. H. García, P. Ezquerro, R. T. Jover, P. Teatini, Y. Li, **2024**. Modeling the Optimal Management of Land Subsidence Due to Aquifers Overexploitation. *Journal of Environmental Management*, 349: 119333. https://doi.org/10.1016/j.jenvman.2023.119333.
- 213. Abhijit D., A., J. Roy, S. Chakraborti. **2016**. Socio-Economic Analysis of Arsenic Contamination of Groundwater in West Bengal. Berlin: Springer.
- 214. Shaji, E., M. Santosh, K. V. Sarath, P. Prakash, V. Deepchand, and B. V. Divyaa. **2020**. Arsenic contamination of groundwater: A global synopsis with focus on the Indian Peninsula. *Geoscience Frontiers*, 12(3): 101079. https://doi.org/10.1016/j.gsf.2020.08.015.

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.